

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Basteck, Christian; Ehlers, Lars

# Conference Paper On (constrained) Efficiency of strategy-proof random assignment

Beiträge zur Jahrestagung des Vereins für Socialpolitik 2024: Upcoming Labor Market Challenges

# Provided in Cooperation with:

Verein für Socialpolitik / German Economic Association

*Suggested Citation:* Basteck, Christian; Ehlers, Lars (2024) : On (constrained) Efficiency of strategyproof random assignment, Beiträge zur Jahrestagung des Vereins für Socialpolitik 2024: Upcoming Labor Market Challenges, ZBW - Leibniz Information Centre for Economics, Kiel, Hamburg

This Version is available at: https://hdl.handle.net/10419/302429

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

# On (constrained) efficiency of strategy-proof random assignment<sup>\*</sup>

Christian Basteck<sup> $\dagger$ </sup> Lars Ehlers<sup> $\ddagger$ </sup>

February 14, 2024

#### Abstract

We study random assignment of indivisible objects among a set of agents, when each agent is to receive one object and has strict preferences over the objects. Random Serial Dictatorship (RSD) satisfies equal treatment of equals, ex-post efficiency and strategy-proofness. Answering a longstanding open question, we show that RSD is not characterized by those properties – there are other mechanisms satisfying equal treatment of equals, ex-post efficiency and strategy-proofness which are not welfare-equivalent to RSD. On the other hand, we show that RSD is not Pareto-dominated by any mechanism that is (i) strategy-proof and (ii) boundedly invariant. Moreover, the same holds for all mechanisms that are ex-post efficient, strategy-proof and boundedly invariant: no such mechanism is dominated by any other mechanism that is strategy-proof and boundedly invariant.

JEL Classification: D63, D70.

*Keywords:* random assignment, strategy-proofness, ex-post efficiency, bounded invariance.

<sup>\*</sup>The first author gratefully acknowledges support by the Deutsche Forschungsgemeinschaft through CRC TRR 190 (project number 280092119) while the second author acknowledges financial support from the SSHRC (Canada) under Insight Grant 435-2023-0129. We thank William Thomson for useful comments and suggestions on an earlier version.

<sup>&</sup>lt;sup>†</sup>WZB Berlin, Berlin, Germany. Email: christian.basteck@wzb.eu.

<sup>&</sup>lt;sup>‡</sup>Département de Sciences Économiques and CIREQ, Université de Montréal, Montréal, QC H3C 3J7, Canada. Email: lars.ehlers@umontreal.ca.

# 1 Introduction

Consider the problem of assigning indivisible objects among a set of agents – each agent is to receive one object and has strict preferences over the set of objects. Further, while objects' characteristics may include a fixed monetary payment, there are no additional transfers. Problems like this arise in many real-life situations such as the assignment of on-campus housing (where rents are fixed), organ allocation, school choice with ties in applicants' priorities, etc. Whenever several agents prefer the same object over any other, the indivisible nature of objects, together with the absence of compensating transfers, will render any deterministic assignment unfair. For that reason, both theorists and policy makers have turned to random assignments in such contexts.

To implement random assignments, a mechanism will have to elicit agents' preferences to then determine a probability distribution over deterministic assignments. Since eliciting preferences over all possible lotteries is often impractical, agents are typically only asked to report their preference ranking over objects – for example, school choice programs will typically ask applicants to provide a list of schools, ranked from most- to least-preferred. Crucially, given that preferences are private information, the design of random assignment mechanisms has to take into account agents' incentives to reveal their preferences.

Strategy-proofness makes truthful reporting a dominant strategy and thus should ensure that agents truthfully reveal their ordinal preferences over objects for any underlying utility representation of preferences. Unfortunately, the literature on random assignment mechanisms contains numerous impossibility results as soon as strategy-proofness and equaltreatment-of-equals, as a minimal fairness requirement, are married with different ex-ante notions of efficiency.<sup>1</sup> Hence we will focus on ex-post efficiency and analyse *constrained* exante efficiency in a class of mechanisms satisfying certain properties. Furthermore, we will consider situations where each agent needs to be assigned exactly one object. We will refer to this as the acceptable domain, as agents either desire all objects (but cannot consume more than one) or cannot unilaterally reject an assignment.

One of the most prominent procedures, frequently used in real life, is random serial dictatorship (RSD): After ordering agents uniformly at random, the first agent gets to pick their most preferred object and each subsequent agent gets to pick their most preferred among all remaining objects. Besides being easily implementable, RSD satisfies many desirable properties: (1) equal treatment of equals – any two agents with the same preferences obtain identical random assignments ex-ante, (2) ex-post efficiency – for any realized ordering of

<sup>&</sup>lt;sup>1</sup>Throughout 'ex-ante' is to be understood as before realizing the final deterministic assignment; this corresponds to the term 'interim' used in mechanism design outside of the literature on random assignments.

agents, the associated deterministic assignment is Pareto efficient, and (3) strategy-proofness – no agent has an incentive to pick a less preferred object when it is their turn to choose. Moreover, the procedure can be readily implemented as a direct mechanism where agents report their ordering over objects and the procedure picks optimally on their behalf. In that case strategy-proofness is satisfied in that it is a dominant strategy to report preferences truthfully.

It has been an open question for more than two decades whether RSD is characterized by these properties in terms of welfare, i.e., whether any mechanism satisfying (1)-(3) necessarily yields the same individual random assignments as RSD.<sup>2</sup> Our first main result invalidates this conjecture—there exist mechanisms which satisfy equal treatment of equals, ex-post efficiency and strategy-proofness, and for which for some preference profile and some agent, her random assignment does not coincide with the one of RSD. In fact for some preference profiles our constructed mechanism yields random assignments that Pareto dominate, in a stochastic dominance sense, the ones arising under RSD. Hence, as preferences over objects are strict, for any extension of preferences from objects to random assignments, all agent prefer this random assignment (with strict preference holding for some agents). Thus, the mechanism is not welfare equivalent to RSD.

In the mechanism constructed for our first main result, the random assignment of a given object may depend on agents' preferences over less preferred objects. This is disturbing as strategy-proofness implies that an individual agent's probability share for a given object *does not* depend on their own preferences over less preferred objects. In contrast, RSD satisfies (4) bounded invariance according to which the random assignment of any object x depends only on agents' preferences over objects which are preferred to x – changing the reported ordering of less preferred objects does not affect the probability with which other agents are assigned an object. Hence, for strategy-proof mechanisms, bounded invariance may be viewed as a weak, object-wise, non-bossiness condition.

To the best of our knowledge, except for Deferred Acceptance (DA), all other mechanisms considered in the literature as well as real-life mechanism used in practice satisfy bounded invariance. For example, Probabilistic Serial, Immediate Acceptance and as well as the Top-Trading-Cycles (TTC) mechanism are all boundedly invariant. Moreover, we show that DA satisfies bounded invariance if and only if the underlying deterministic priority structure ensures Pareto efficiency, adding to the main theorem by Ergin (2002) – hence DA's main weakness, the fact that it may lead to inefficient assignments, can be directly traced back to instances where it violates bounded invariance. Further, we show that bounded invariance plays a key role in another seminal contribution for deterministic mechanisms:

 $<sup>^{2}</sup>$ Bogomolnaia and Moulin (2001) were able to prove this for the case of three agents and three objects.

in the characterization of hierarchial exchange rules by Pápai (2000) reallocation-proofness may be replaced by bounded invariance.

Our second main result is that no mechanism satisfying properties (2)-(4) is Pareto dominated (in terms of first-order stochastic dominance) by a strategy-proof and boundedly invariant mechanism. As an immediate corollary we find that RSD is not Pareto dominated by any mechanism satisfying strategy-proofness and bounded invariance. It is important to stress that our main result applies to any mechanism and is not exclusive to RSD. For instance, in applications one might take into account affirmative action constraints with respect to minorities or disadvantaged groups by not choosing certain orders of agents (where majorities or advantaged groups come first in the order) and apply a weighted version of RSD.<sup>3</sup> Any such mechanism satisfies (2)-(4) and is therefore not Pareto dominated by any strategy-proof and boundedly invariant mechanism. This addresses a question whether RSD is constrained efficient in the class of strategy-proof mechanisms (where we impose bounded invariance in addition), and provides a positive partial answer on the acceptable domain to the long-standing open question by Zhou (1990) whether RSD is undominated in the class of mechanisms satisfying (1)-(3) – our result does not impose (1) but instead imposes (4). This is the first affirmative result for RSD in connection with ex-post efficiency and strategy-proofness.

We connect our main results to the previous literature. Numerous contributions establish the impossibility of strategy-proofness, envy-freeness and ex-ante efficiency. In the cardinal framework, Zhou (1990) showed that no mechanism satisfies equal treatment of equals, strategy-proofness and ex-ante efficiency (where the latter postulates always to choose a random assignment which is not Pareto dominated in terms of expected utilities by any other one). In the ordinal framework, Bogomolnaia and Moulin (2001) establish an analogous impossibility result where envy-freeness is weakened to equal-treatment-of-equals. Nesterov (2017) shows that the impossibility persists when ex-ante efficiency is weakened to ex-post efficiency (while maintaining envy-freeness).<sup>4</sup> Shende and Purohit (2023) show that strategy-proofness and envy-freeness are incompatible with unanimity<sup>5</sup> (which they refer to as contention-free efficiency), a significant weakening of ex-post efficiency. Further, Basteck and Ehlers (2023) show that a strategy-proof and envy-free mechanism is ex-post unanimous with probability of at most  $\frac{2}{n}$  (where *n* is the number of agents). In other words, for

 $<sup>^{3}</sup>$ For example, if there are several subgroups of agents, we may wish to randomize over orders in which members of the groups alternate when 'picking' objects.

<sup>&</sup>lt;sup>4</sup>Zhang (2019) proves a strong group-manipulability result, imposing ex-post efficiency and auxiliary fairness axioms that are by themselves weaker than envy-freeness.

<sup>&</sup>lt;sup>5</sup>Unanimity requires that whenever all agents consider a different object most-preferred, each should receive their most-preferred object. In other words, whenever there is a unique Pareto efficient assignment, it is chosen with probability one.

any strategy-proof and envy-free mechanism there exist preference profiles where the unique ex-post efficient assignment is chosen with probability of at most  $\frac{2}{n}$  (and inefficient assignments are chosen with probability of at least  $1 - \frac{2}{n}$ ). This finding strengthens significantly the incompatibility of strategy-proofness, envy-freeness and ex-post efficiency and provides an exact upper bound for ex-post unanimity. Bogomolnaia and Moulin (2001) introduced the probabilistic serial (PS) mechanism and show that it is envy-free and ex-ante efficient (hence necessarily violates strategy-proofness). Bogomolnaia and Heo (2012) provide an axiomatic characterization of the PS mechanism via ex-ante efficiency, envy-freeness and bounded invariance.<sup>6</sup>

Furthermore, one might allow agents to rank certain objects unacceptable. If agents may rank objects unacceptable and possibly receive no object, notions of efficiency have to take into account the set of (un)assigned objects: a deterministic assignment is non-wasteful if no agent prefers an unassigned object to her assignment. As a stronger requirement, ex-ante non-wastefulness demands that if an agent prefers an object over another and is assigned the less-preferred with positive probability, then the more-preferred object must be assigned with probability one. Erdil (2014) established that there are mechanisms Paretodominating RSD which are less exante wasteful, which is a negative answer on the full domain to a question first raised by Zhou (1990). Notably, the mechanism constructed in Erdil (2014, Proposition 3) coincides with RSD on the acceptable domain, i.e., it does not Pareto-dominate RSD for the domain where all objects are acceptable. His constructed mechanism satisfies equal treatment of equals and strategy-proofness but violates bounded invariance. Our second main result implies that any strategy-proof and boundedly invariant mechanism, which dominates RSD on the full domain, must coincide with RSD on the acceptable domain. In other words, Pareto improvements over RSD are only possible for profiles where objects are classified unacceptable in a "certain" way.

The paper is organized as follows. Section 2 introduces random assignments, their properties and several prominent mechanisms. Section 3 constructs a mechanism which satisfies equal treatment of equals, ex-post efficiency and strategy-proofness, and which is not welfareequivalent to RSD. Section 4 states our second main result pertaining to the constrained efficiency of any mechanism satisfying ex-post efficiency, strategy-proofness and bounded invariance. Section 5 considers bounded invariance for deterministic mechanisms in Pápai (2000) and Ergin (2002). Section 6 provides implications of our second main result for tiebreaking of weak priorities. Section 7 concludes. The Appendix contains the proofs of our main results.

<sup>&</sup>lt;sup>6</sup>Hashimoto et al. (2014) weakened bounded invariance to weak invariance in this characterization, a property which is satisfied by any strategy-proof mechanism.

# 2 Model

Let  $N = \{1, \ldots, n\}$  denote the set of agents and  $O = \{o_1, \ldots, o_n\}$  denote the finite set of objects. Throughout the main text we suppose  $|N| = |O| \ge 3$  and allow for unequal numbers of agents and objects in the Appendix. Each agent *i* has strict preferences over  $O \cup \{i\}$  where *i* stands for being unassigned; let  $R_i$  denote the corresponding linear order<sup>7</sup> and write  $P_i$  for its asymmetric part (where  $xP_iy$  is defined by  $xR_iy$  and  $x \ne y$ ). Let  $\mathcal{R}^i$  denote the set of all strict preferences of agent *i* over  $O \cup \{i\}$  such that  $oR_ii$  for all  $o \in O$ , i.e., where all objects are acceptable. Let  $\mathcal{R}^N = \times_{i \in N} \mathcal{R}^i$  denote the set of all preference profiles  $R = (R_1, \ldots, R_n)$ , which we call the acceptable domain.

An assignment is a mapping  $\mu : N \to O \cup N$  such that<sup>8</sup>  $\mu_i \in O \cup \{i\}$  for all  $i \in N$  and  $\mu_i \neq \mu_j$  for all  $i \neq j$ . Let  $\mathcal{M}$  denote the set of all assignments.

An assignment  $\mu$  is efficient under R if there exists no  $\mu' \in \mathcal{M}$  such that  $\mu'_i R_i \mu_i$  for all  $i \in N$  and  $\mu'_j P_j \mu_j$  for some  $j \in N$ . As all objects are acceptable and |O| = |N|, this implies that no agent is unassigned under  $\mu$ . Let  $\mathcal{PO}(R)$  denote the set of all efficient assignments under R. An assignment  $\mu$  is weakly efficient under R if there exists no  $\mu' \in \mathcal{M}$  such that  $\mu'_i P_i \mu_i$  for all  $i \in N$ . Let  $\mathcal{WPO}(R)$  denote the set of all weakly efficient assignments under R.

Let  $\Delta(\mathcal{M})$  denote the set of all probability distributions over  $\mathcal{M}$ . Given  $p \in \Delta(\mathcal{M})$ , let  $p_{ia}$  denote the associated probability of *i* being assigned *a* and refer to  $p_i = (p_{ia})_{a \in O \cup \{i\}}$  as agent *i*'s (individual) random assignment. Let supp(p) denote the support of *p*. Then (i) *p* is ex-post efficient under *R* if  $supp(p) \subseteq \mathcal{PO}(R)$ , and (ii) *p* is ex-post weakly efficient under *R* if  $supp(p) \subseteq \mathcal{WPO}(R)$ .

For all  $i \in N$ , all  $R_i \in \mathcal{R}^i$  and all  $x \in O \cup \{i\}$ , let  $B(x, R_i) = \{y \in O \cup \{i\} : yR_ix\}$ . Then given any  $p, q \in \Delta(\mathcal{M}), p_i$  stochastically  $R_i$ -dominates  $q_i$  if for all  $x \in O \cup \{i\}$ ,

$$\sum_{y \in B(x,R_i)} p_{iy} \ge \sum_{y \in B(x,R_i)} q_{iy}.$$

A random assignment p stochastically R-dominates (or sd-dominates) another random assignment q if  $p_i R_i$ -dominates  $q_i$  for all  $i \in N$ . A random assignment is stochastic dominance (sd)-efficient if there is no random assignment  $q \neq p$  that stochastically R-dominates it.<sup>9</sup> Given two random assignments p and q, we say that p and q are welfare-equivalent if  $p_i = q_i$ 

<sup>&</sup>lt;sup>7</sup>Thus,  $R_i$  is (i) complete, (ii) transitive and (iii) antisymmetric  $(xR_iy \text{ and } yR_ix \text{ implies } x = y)$ .

<sup>&</sup>lt;sup>8</sup>We will use throughout the convention to write  $\mu_i$  instead of  $\mu(i)$  for any  $i \in N$ .

<sup>&</sup>lt;sup>9</sup>Bogomolnaia and Moulin (2001) refer to this as "ordinal efficiency". It implies Pareto-efficiency with respect to expected utilities for some von Neumann-Morgenstern-representations of agents' ordinal preferences over objects (McLennan, 2002).

for all  $i \in N$ .<sup>10</sup>

A mechanism (or rule) is a mapping  $f : \mathbb{R}^N \to \Delta(\mathcal{M})$ . Then f(R) denotes the random assignment chosen for R, and  $f_{ia}(R)$  denotes the probability of agent i being assigned object a. For  $i \in N$ ,  $f_i(R)$  denotes the tuple of assignment probabilities  $(f_{ia}(R))_{a \in O}$ , and for  $a \in O$ ,  $f_a(R)$  is defined accordingly as the tuple of probability shares with which a is assigned to the various agents. A mechanism f sd-dominates another mechanism g, denoted as  $f \triangleright^{sd} g$ , if for any profile R the random assignment f(R) stochastically R-dominates the random assignment g(R), and for some profile  $\overline{R}$  and  $i \in N$  we have  $f_i(\overline{R}) \neq g_i(\overline{R})$ . Further fis sd-efficient if for all  $R \in \mathcal{R}^N$ , f(R) is sd-efficient under R. Similarly, we define ex-post (weak) efficiency for a mechanism. A mechanism f is deterministic if for any profile R, |supp(f(R))| = 1, i.e. the mechanism chooses one assignment with probability one.

Then f is strategy-proof if for all  $R \in \mathbb{R}^N$ , all  $i \in N$  and all  $R'_i \in \mathbb{R}^i$ ,  $f_i(R)$  stochastically  $R_i$ -dominates  $f_i(R'_i, R_{-i})$ . Strategy-proofness is equivalent to the requirement that for any von Neumann-Morgenstern utility presentation compatible with a given ordinal ranking of objects, submitting the true ordinal ranking maximizes an agent's expected utility. Most real-life mechanisms only elicit this ordinal information (instead of von Neumann-Morgenstern utilities).

Furthermore, f is envy-free if for all  $R \in \mathbb{R}^N$  and all  $i \in N$ ,  $f_i(R)$  stochastically  $R_i$ dominates  $f_j(R)$  (where in  $f_j(R)$  the outside option j is replaced by i). If f(R) attaches probability one to assignment  $\mu$ , then this is equivalent to  $\mu_i R_i \mu_j$  for all  $i, j \in N$ . Finally, fsatisfies symmetry (or more descriptively, equal treatment of equals) if for all  $R \in \mathbb{R}^N$  and all  $i, j \in N$ ,  $R_i = R_j$  implies  $f_{io}(R) = f_{jo}(R)$  for all  $o \in O$ .

Most properties are defined in terms of an agents' random assignments. For a given set of properties, we say that a mechanism f is unique in terms of probability shares, if for any other mechanism  $\phi$  satisfying this set of properties, f(R) and  $\phi(R)$  are welfare-equivalent for any profile R, i.e., if individual random assignments coincide. Below we introduce two well-known mechanisms.

Let  $\succ$  denote a strict priority ranking over N and let  $\Pi$  denote the set of all strict priority orders. Given  $\succ \in \Pi$ , let  $f^{\succ}$  denote the (deterministic) serial dictatorship (SD) mechanism where agents are assigned their most-preferred among all available objects in

<sup>&</sup>lt;sup>10</sup>Some papers directly define a bistochastic matrix  $(p_{ia})_{i \in N, a \in O}$  rather than a random assignment per se, i.e., a convex combination of deterministic assignments. Nonetheless, corresponding random assignments exist as any bistochastic matrix  $(p_{ia})_{i \in N, a \in O}$  can be decomposed as a convex combination of deterministic assignments by the Birkhoff-von Neumann Theorem (Birkhoff, 1946). Abdulkadiroğlu and Sönmez (2003a) observe that an ex-post efficient random assignment may be welfare-equivalent to a random assignment with support contained in the set of inefficient assignments so that  $p_i = q_i$  for all  $i \in N$  does not imply p = q. The latter also observed by Pycia and Troyan (2023) for RSD.

order of their priority.<sup>11</sup> Then the random serial dictatorship (RSD) mechanism is defined by  $RSD(R) = \frac{1}{n!} \sum_{\kappa \in \Pi} f^{\kappa}(R)$  for all  $R \in \mathcal{R}^N$ .

We omit the formal definition of the probabilistic serial (PS) mechanism<sup>12</sup> and provide an intuitive formulation instead: agents start eating, with uniform speed, from their mostpreferred object; once an object is exhausted, each agent eats with uniform speed from his most-preferred among the remaining objects, and so on until all objects are exhausted. The assignment probabilities of any agent in PS are simply the shares of objects the agent has eaten over the course of this process.<sup>13</sup>

The literature widely discusses the tradeoff among these two mechanisms: on the one hand RSD satisfies ex-post efficiency, symmetry and strategy-proofness but violates sdefficiency and envy-freeness while on the other hand PS satisfies sd-efficiency and envyfreeness but violates strategy-proofness.

## 3 Ex-post Efficiency, Symmetry and Strategy-proofness

It has long been an open question, at least since Bogomolnaia and Moulin (2001) were able to prove the statement for |N| = |O| = 3, whether random serial dictatorship is characterized (in terms of welfare) by ex-post efficiency, symmetry and strategy-proofness. As we show, this is not the case for five agents or more.

**Theorem 1** For five agents or more, there exist mechanisms satisfying ex-post efficiency, symmetry and strategy-proofness, which are not welfare-equivalent to random serial dictator-ship.

We give an informal description of the main steps of the construction of such a mechanism for five agents and five objects below. The starting point of the construction is inspired by Erdil (2014, Proposition 3), and adapted to the acceptable domain in an inventive way.<sup>14</sup> The detailed demonstration is relegated to the Appendix.

 $<sup>\</sup>frac{1}{11} \text{For any } R \in \mathcal{R}^N \text{ and } i_1 \succ i_2 \succ \cdots \succ i_n, i_1 \text{ receives his most } R_{i_1} \text{-preferred object in } O \text{ (denoted by } f_{i_1}^{\succ}(R)\text{), and for } l = 2, \ldots, n, i_l \text{ receives his most } R_{i_l} \text{-preferred object in } O \setminus \{f_{i_l}^{\succ}(R), \ldots, f_{i_{l-l}}^{\succ}(R)\} \text{ (denoted by } f_{i_l}^{\succ}(R)\text{).}$ 

<sup>&</sup>lt;sup>12</sup>For that, we refer the reader to Bogomolnaia and Moulin (2001); Bogomolnaia (2015) offers an alternative definition of PS, and Katta and Sethuraman (2006) extend PS to the domain where indifferences are allowed.

<sup>&</sup>lt;sup>13</sup>The PS-mechanism pins down individuals' object assignment probabilities directly but can be decomposed as a convex combination of deterministic assignments by the Birkhoff-von Neumann Theorem (Birkhoff, 1946).

<sup>&</sup>lt;sup>14</sup>Erdil (2014) considers random assignment with *unacceptable* objects where RSD may leave some objects unassigned and shows that it is possible to assign them with higher probability without violating strategy-proofness. Note that on the acceptable domain, his constructed mechanism coincides with RSD.

First, we describe an alternative formulation of RSD. Namely, define the mechanism  $g^i$  where the four agents in  $N \setminus \{i\}$  get to choose in random order like in RSD, while *i* is assigned the residual object. Now randomizing over all such mechanisms  $g^i$ ,  $i \in N$ , with probability 1/5 gives us back RSD.

Second, we describe a mechanism which strictly sd-dominates  $g^{5}$  for agent 1 and yields the same random assignments for agents 2, 3 and 4. Let  $R_{2}, R_{3}, R_{4}$  be as follows

$R_2$	$R_3$	$R_4$
c	С	с
a	b	e
d	d	d
e	e	a
b	a	b

Objects d and e will play the role of leftover objects where it will be crucial for agents 2 and 3 to rank the same leftover object d above e and for agent 4 to rank e above d. Now for  $g^{5}$ , agent 1 does not get a, b or c for the orders 4 - 2 - 3 - 1 - 5 and 4 - 3 - 2 - 1 - 5 (where any order has probability 1/24), i.e. at least with probability 1/12.

Now suppose  $R_1 : ab \dots^{15}$  Note that agent 5 gets object *b* for the orders 3-1-2-4-5 and 3-1-4-2-5, i.e. with at least probability 1/12. We will increase for agent 1 the share of *b* by 1/12 (while keeping his share of *a* unchanged) and reduce for agent 5 the share of *b* by 1/12, i.e. agent 1 will receive (18a + 6b)/24 in the new mechanism  $g^{1-5}$ . Agents 2, 3 and 4 always get the same random assignment under  $g^{1-5}$  and  $g^5$ , and agent 5's random assignment is the residual.

We verify that ex-post efficiency for  $g^{1-5}$  is preserved. This can be done by replacing the efficient assignments on the left below with the efficient assignments on the right with probability share 1/12 each. Let  $\{d, e\} = \{x, y\}$  and  $xP_1y$ :

$$\begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ x & a & b & c & y \end{pmatrix} + \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ a & d & c & e & b \end{pmatrix} \rightarrow \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ b & a & c & e & d \end{pmatrix} + \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ a & d & b & c & e \end{pmatrix}$$

Note that the first assignment is obtained for the orders 4-2-3-1-5 and 4-3-2-1-5, the second one for the orders 3-1-2-4-5 and 3-1-4-2-5 (since agents 2 and 4 have opposite preferences over d and e), the third one for the orders 3-2-1-4-5 and 3-2-4-1-5, and the fourth one for the orders 4-1-2-3-5 and 4-1-3-2-5.

Indeed, we thus improve agent 1's share of b by 1/12 whenever agent 1 prefers a over b

<sup>&</sup>lt;sup>15</sup>We use this notation to write  $aP_1bP_1o$  for all  $o \in O \setminus \{a, b\}$ .

and prefers both over d and e. In an analogous way, we improve agent 1's share of a by 1/12 whenever agent 1 prefers b over a and prefers both over over d and e. Otherwise, whenever the preferences of agents 2, 3 and 4 are not as above or agent 1 prefers d or e over a or b, let  $g^{1-\underline{5}}(R)$  and  $g^{\underline{5}}(R)$  coincide.

One can verify strategy-proofness of  $q^{1-5}$ . Obviously, agent 1 and 5 are treated differently. In order to recover symmetry, we first permute the roles of agents 1, 2, 3 and 4 and obtain a mechanism which treats those agents equally, and finally we may choose arbitrarily an agent to be the last one to choose.<sup>16</sup> In doing this for  $g^{5}$  we get back RSD but not for  $g^{1-5}$ . For instance, for  $R_1$ : abecd and  $R_5$ : edcab and  $(R_2, R_3, R_4)$  as above,  $g^{\underline{5}}$  gives a positive share of e to agent 1 whereas  $q^{1-5}$  gives zero share of e to agent 1 (as his share of b is increased by 1/12). Permuting the roles of agents and considering  $q^{i-j}(R)$  and  $q^{j}(R)$ , we find that for i = 5, j = 1 the two coincide as 5 does not rank a and b above d and e (as 1 did in  $q^{1-5}(R)$ ) while for  $\{i, j\} \neq \{1, 5\}$  we find that  $g^{i-j}(R)$  and  $g^{j}(R)$  coincide as either 1 or 5 is now in the role of 2, 3, or 4 in  $g^{1-\underline{5}}(R)$  but does not rank c at the top. Considering the convex combination of the permuted mechanisms agents 2, 3 and 4 get the same random assignment and agent 5 is better off for the profile R and  $q^{1-5}$  (as his share of b is decreased by 1/12and otherwise she gets the same random assignment. Thus, the constructed mechanism and RSD are not welfare-equivalent as the random assignment for the constructed mechanism stochastically R-dominates the one of RSD for all agents, i.e. agents 1 and 5 are ex-ante better off while agents 2, 3 and 4 are indifferent.<sup>17</sup>

Furthermore, when there are more than five agents, the first four agents play the same role as above and we let the remaining agents choose in a fixed order, and obtain the same conclusions.

One can even recover neutrality by permuting the names of the objects. Hence, Theorem 1 remains true when neutrality is added. We also show that for this conclusion it suffices to have at least five agents and at least five objects (but possibly with unequal numbers of agents and objects). We establish all this in the Appendix.

We finish with the observation that in the above construction, we may ignore agent 5 and consider assigning five objects among four agents. Then the constructed mechanism sd-dominates RSD, i.e. even though under RSD any agent is always assigned an object (or RSD has size one), RSD might be sd-dominated when there are more objects than agents. Erdil (2014) has studied in detail random assignment with outside options, i.e. where agents may be unassigned and may prefer this to certain objects. He showed that RSD may be

<sup>&</sup>lt;sup>16</sup>In fact, by permuting the roles of agents, we even ensure anonymity, see Appendix A.

<sup>&</sup>lt;sup>17</sup>At the same time (i) there exists no profile for which RSD sd-dominates the constructed mechanism, and (ii) agent 5 might be worse off for the constructed mechanism for the profile  $(R_{-5}, R'_5)$  with  $R'_5$ : bedca.

sd-dominated where his constructed mechanism coincides with RSD when agents find all objects acceptable, i.e. on the acceptable domain.

## 4 Non-Domination

In the constructed mechanism above, the random assignment of a certain object may depend on preferences over less preferred objects.<sup>18</sup> Hence, even though strategy-proofness ensures that an agent's probability share for a particular object is unaffected by changes to the order in which they rank less preferred objects, such changes may still affect the probability shares of this object for *other* agents. The following invariance condition, called bounded invariance, rules out such effects and may therefore be interpreted as a weak object-wise non-bossiness condition for strategy-proof mechanisms. It was first proposed by Bogomolnaia and Heo (2012) who used it in conjunction with ex-ante efficiency and envy-freeness to characterize the probabilistic serial mechanism.<sup>19</sup>

**Definition 1** Given  $i \in N$ ,  $R_i \in \mathcal{R}^i$  and  $x \in O$ , let  $R_i(x) = R_i|B(x, R_i)$  denote the restriction of  $R_i$  to the weak upper contour set of x. Now a mechanism f satisfies bounded invariance (BI) if for all  $R \in \mathcal{R}^N$ , all  $i \in N$ , all  $R'_i \in \mathcal{R}^i$  and all  $x \in O$ , if  $R'_i(x) = R_i(x)$ , then  $f_x(R) = f_x(R'_i, R_{-i})$ .

In other words, if agent *i*'s preference above object x remains unchanged, then the random assignment of x remains unchanged.

Recall that a mechanism f sd-dominates another mechanism g if for any profile R the random assignment f(R) stochastically R-dominates the random assignment g(R), and for some profile  $\bar{R}$  and  $i \in N$  we have  $f_i(\bar{R}) \neq g_i(\bar{R})$ .

**Theorem 2** On the acceptable domain, if a mechanism g satisfies ex-post efficiency, bounded invariance and strategy-proofness, then no boundedly invariant and strategy-proof mechanism sd-dominates g.

RSD satisfies ex-post efficiency, bounded invariance and strategy-proofness – hence, by Theorem 2, RSD is not sd-dominated by any mechanism satisfying bounded invariance and

<sup>&</sup>lt;sup>18</sup>For instance, for the above profile R agent 1 receives 1/12 more of object b under  $g_1^{1-\underline{5}}(R)$  compared to  $g_1^{\underline{5}}(R)$ , i.e.  $g_{1b}^{1-\underline{5}}(R) \neq g_{\underline{5}b}(R)$ . Now if we change  $R_3$  to  $R'_3$ : cba..., then for  $R' = (R'_3, R_{-3})$ , we have  $g_1^{1-\underline{5}}(R') = g_1^{\underline{5}}(R')$ . Then for the mechanism where we permute the roles of agents as above, call it h, we have  $h_{1b}(R) \neq RSD_{1b}(R)$  but  $h_{1b}(R') = RSD_{1b}(R')$  (even though agent 3 kept unchanged his two most preferred objects, i.e.  $R_3: cb...$  and  $R'_3: cb...$ ).

<sup>&</sup>lt;sup>19</sup>The set of all ex-ante efficient, strategy-proof, non-bossy, neutral, and boundedly invariant mechanisms has recently been characterized by Alva et al. (2024).

strategy-proofness. The same is true for weighted versions of RSD, i.e., where we attach different weights to different orders of agents and apply SD. Such weights could take into account minorities/majorities and (dis)advanted groups.<sup>20</sup> Furthermore, in Theorem 2 expost efficiency cannot be weakened to ex-post weak efficiency. For instance, the Random-Dictatorship-cum-Equal-Division<sup>21</sup> by Basteck and Ehlers (2023) satisfies ex-post weak efficiency, bounded invariance and strategy-proofness, but is sd-dominated by RSD.

Several questions remain. First, does Theorem 2 remain true when we drop bounded invariance as a requirement on the second mechanism, i.e., keep bounded invariance only for the first mechanism, thus strengthening the implication? Second, could we drop bounded invariance as a requirement on the first mechanism, weakening Theorem 2's premise? Third, is RSD characterized by ex-post efficiency, bounded invariance, strategy-proofness and symmetry?

We provide an outline of the proof of Theorem 2. As a basic step we show that for any efficient deterministic assignment, any agent must rank his allocated object weakly above some non-top ranked object. Then for a fixed object, say z, we count for any profile and for any agent the number of non-top ranked objects below z, and consider lexicographic minimization with respect to those numbers. If g sd-dominates f, then the set of profiles where f and g differ is non-empty. Now in this set we choose a profile where object z is ranked as low as possible with respect to the minimization outlined above and show that the random assignment of z must coincide for f and g. Remaining in the set of profiles where f and g differ and z is ranked as low as possible, we take another object, say y, choose a profile where y is ranked as low as possible and show that the random assignment of y (and z) is identical for f and g. Iterating we eventually exhaust the set of objects and obtain that f and g coincide, which implies that the set of profiles where f and g differ was empty yielding the final contradiction.

### 5 Bounded Invariance for Deterministic Mechanisms

Even though bounded invariance has been introduced for random assignment and has not been explicitly considered for deterministic assignment, we show below that it plays a crucial role for two seminal contributions in the literature.

 $<sup>^{20}</sup>$ E.g., if objects are to be assigned to a set of agents comprised of two groups, one might consider orders that alternate between members of the two subgroups and randomize within each.

<sup>&</sup>lt;sup>21</sup>We omit the formal definition and refer to Basteck and Ehlers (2023). Informally, the mechanism works as follows: any agent *i* is chosen with probability  $\frac{1}{n}$ , then agent *i* picks his most preferred object and the remaining objects are assigned uniformly among the other agents.

#### 5.1 Hierarchical Exchange

Pápai (2000) introduces and characterizes hierarchical exchange rules. Extreme cases are the top-trading-cycles (TTC) rule where each agent owns one house (or object)<sup>22</sup> and the SD rule where houses are allocated in a certain order of agents. For the TTC rule each agent points to his most preferred house and each house points to its owner, before trading cycles are then executed iteratively. For the SD rule, the first agent to choose implicitly owns all houses and picks his most preferred house, then the second agent implicitly inherits all unassigned houses from the first agents and picks his most preferred house, and so on. Hierarchical exchange rules give a precise formulation of those ideas by covering immediate cases as follows: any house is owned by one agent, any agent points to his most preferred house, and each house points to its owner. Then following the execution of trading cycles, unassigned objects are inherited by agents who remain unassigned, and so on. We give a formulation of hierarchical exchange rules based on Pycia and Ünver (2017).

A submatching on  $I \subseteq N$  is an injective function  $\sigma : I \to O$ . A submatching allocates a subset of houses to a subset of agents so that no two different agents obtain the same house, i.e.  $\sigma(i)$  is the house matched with agent  $i \in N$  and  $\sigma^{-1}(h)$  is the agent matched with house  $h \in \sigma(I)$ . The set of submatchings is denoted by  $\mathcal{S}$ . For each  $\sigma \in \mathcal{S}$ , the set of matched agents is denoted by  $I_{\sigma}$ , and the set of matched houses is denoted by  $H_{\sigma}$ . We write  $\overline{I}_{\sigma}$  for  $N \setminus I_{\sigma}$  and  $\overline{H}_{\sigma}$  for  $O \setminus H_{\sigma}$ .

**Definition 2** An ownership structure is a collection of mappings

$$(c_{\sigma}:\overline{H}_{\sigma}\to\overline{I}_{\sigma})_{\sigma\in\mathcal{S}}$$

where at submatching  $\sigma$ , the function  $c_{\sigma}$  maps each unmatched house h to the unique unmatched agent  $c_{\sigma}(h)$  who owns it then, and the following (consistency) condition is satisfied: for any  $\sigma, \sigma' \in S$  such that  $\sigma \subseteq \sigma'$ , if  $i \in \overline{I}_{\sigma'}$  owns house  $h \in \overline{H}_{\sigma'}$  at  $\sigma$  ( $c_{\sigma}(h) = i$ ), then i owns h at  $\sigma'$  ( $c_{\sigma'}(h) = i$ ).

#### Hierarchical Exchange Algorithm:

Given an ownership structure  $(c_{\sigma})_{\sigma \in S}$ , the algorithm starts with empty the submatching  $\sigma^0 = \emptyset$ , and in each round r = 1, 2, ... it matches some agents with houses. Let  $\sigma^{r-1}$  we denote the submatching of agents matched before round r. If  $\sigma^{r-1} \notin \mathcal{M}$ , then the algorithm proceeds with the following steps of round r.

<sup>&</sup>lt;sup>22</sup>Throughout this subsection we often use the term house instead of object.

Step 1: Pointing. Each house  $h \in \overline{H}_{\sigma^{r-1}}$  points to the agent who owns it at  $\sigma^{r-1}$ . Each agent  $i \in \overline{I}_{\sigma^{r-1}}$  points to his most preferred house in  $\overline{H}_{\sigma^{r-1}}$ .

Step 2: Execution of trading cycles. A trading cycle is given by  $h_1 \to i_1 \to \cdots \to h_k \to i_k \to h_1$  in which  $k \in \{1, 2, \ldots\}$ , agent  $i_l \in \overline{I}_{\sigma^{r-1}}$  points at house  $h_{l+1} \in \overline{H}_{\sigma^{r-1}}$ , and house  $h_l$  points to  $i_l$ . Each agent in each trading cycle is matched with the house she is pointing to.

Step 3. Submatching  $\sigma^r$  is defined as the union of  $\sigma^{r-1}$  and the set of newly matched agent-house pairs. When all agents or all houses are matched under  $\sigma^r$ , then the algorithm terminates and gives matching  $\sigma^r$  as its outcome. Otherwise we continue with round r + 1.

Now a rule is a hierarchical exchange rule if we can find an ownership structure such that the outcomes of the rule and the above algorithm are identical.

Reallocation-proofness excludes the possibility that two individuals can gain by jointly manipulating the outcome and swapping objects ex-post, when the collusion is self-enforcing in the sense that neither party can lose by reporting false preferences in case the other party does not adhere to the agreement and reports honestly.

**Definition 3** A mechanism f is manipulable via reallocation if there exist R,  $i, j \in N$  and  $R'_i, R'_j$  such that  $f_h(R) = f_h(R'_h, R_{-h}) \neq f_h(R'_i, R'_j, R_{-ij})$  for h = i, j, and both  $f_j(R'_i, R'_j, R_{-ij})P_if_i(R)$  and  $f_i(R'_i, R'_j, R_{-ij})R_jf_j(R)$ . An assignment rule is reallocation-proof if it is non-manipulable via reallocation.

The following property rules out manipulations of other agents' assignments without changing the agent's own assignment. A deterministic mechanism f satisfies non-bossiness if for any profiles R and  $(R'_i, R_{-i})$  (where  $i \in N$ ),  $f_i(R) = f_i(R'_i, R_{-i})$  implies f(R) = $f(R'_i, R_{-i})$ .

We show the following basic implication, namely, strategy-proofness, non-bossiness and bounded invariance imply reallocation-proofness.

**Lemma 1** For deterministic mechanisms, strategy-proofness, non-bossiness and bounded invariance imply reallocation-proofness.

**Proof.** Suppose that f is manipulable via reallocation. Then by non-bossiness, we have  $f(R) = f(R'_h, R_{-h})$  for h = i, j. Thus,  $f(R) = f(R'_i, R_{-i}) = f(R'_j, R_{-j})$ . Let  $a = f_i(R) = f_i(R'_j, R_{-j})$ ,  $b = f_j(R'_i, R'_j, R_{-ij})$  and  $c = f_i(R'_i, R'_j, R_{-ij})$ .

Since f is manipulable via reallocation, we have  $a = f_i(R'_j, R_{-j}) \neq f_i(R'_i, R'_j, R_{-ij}) = c$ . Hence, by strategy-proofness,  $aP_ic$  and  $cP'_ia$ . Furthermore,  $bP_iaP_ic$ .

Let  $R''_i$  be defined as follows:  $R''_i$  ranks first  $B(R_i, a) \setminus \{a\}$  according to  $R_i$ , then c, and then all remaining objects according to  $R_i$ , i.e.  $R''_i : R_i | B(R_i, a) \setminus \{a\}, c, a, \cdots$ . Then by strategy-proofness,  $f_i(R'_i, R'_j, R_{-ij}) = c$  and  $f_i(R'_j, R_{-j}) = a$ , we obtain  $f_i(R''_i, R'_j, R_{-ij}) = c$ . Thus, by non-bossiness,  $f(R'_i, R'_j, R_{-ij}) = f(R''_i, R'_j, R_{-ij})$ .

But then  $f_j(R''_i, R'_j, R_{-ij}) = f_j(R'_i, R'_j, R_{-ij}) = b \neq f_j(R'_j, R_{-j})$  contradicts bounded invariance as  $b \in B(R_i, a) \setminus \{a\} = B(R''_i, c) \setminus \{c\}$  and  $R_i | B(R_i, a) \setminus \{a\} = R''_i | B(R''_i, c) \setminus \{c\}$ .  $\Box$ 

Note that Lemma 1 does not use efficiency but will be crucial for the main result of Pápai (2000). Below we show replacing reallocation-proofness with bounded invariance yields the same characterization of hierarchical exchange rules.

**Theorem 3** The following are equivalent:

- (i) f satisfies strategy-proofness, non-bossiness, efficiency and bounded invariance.
- (ii) f satisfies strategy-proofness, non-bossiness, efficiency and reallocation-proofness.
- (iii) f is a hierarchical exchange rule.

**Proof.** (i) $\Rightarrow$ (ii): This follows from Lemma 1.

 $(ii) \Leftrightarrow (iii)$  follows from Pápai (2000, Theorem).

(iii) $\Rightarrow$ (i): Suppose f is a hierarchical exchange rule. Then by (ii), f satisfies strategyproofness, non-bossiness and efficiency. Suppose f violates bounded invariance. But then we must have  $a = f_i(R) \neq f_i(R'_i, R_{-i}) = b$  (as otherwise we apply non-bossiness to obtain the desired conclusion). Thus, by strategy-proofness,  $aP_ib$  and  $bP'_ia$ . Now for some  $c \in B(R_i, a)$ we have  $f_c(R) \neq f_c(R'_i, R_{-i})$ .

Using strategy-proofness, non-bossiness and  $R_i(c) = R'_i(c)$ , without loss of generality, we may suppose that  $R_i : cab \dots$  and  $R'_i : cba \dots$ 

But then for f under R and  $(R'_i, R_{-i})$ , agent i always points to c and is not allocated any object before object c is allocated. As the pointing and the allocations/inheritances of other agents are independent of i's pointing before i is allocated his object, object c must be assigned to the same agent under R and  $(R'_i, R_{-i})$ .

Pycia and Unver (2017) characterize trading cycles algorithms with ownership and brokerage via efficiency, strategy-proofness and non-bossiness. Loosely speaking a broker owns a house which she is only allocated if nobody else wants it (or in other words she is forbidden to point to his owned house). Now by the above theorem, rules with a broker violate both bounded invariance and reallocation proofness.<sup>23,24</sup>

<sup>&</sup>lt;sup>23</sup>This can also be seen in the usual three agent-three object example where  $N = \{1, 2, 3\}$ ,  $O = \{a, b, c\}$ , 1 owns a, 2 owns b, 3 owns c, and agent 1 is a broker for a, i.e. 1 is not allocated a, and agents 2 and 3 want a. For  $R_1 : abc$ , 1 and 2 exchange objects (i.e. 2 gets a) and for  $R'_1 : acb$ , 1 and 3 exchange objects (i.e. 3 gets a), which is a violation to bounded invariance.

<sup>&</sup>lt;sup>24</sup>One may then wonder whether strategy-proofness, efficiency and bounded invariance imply non-

We stress again that in the mechanism constructed for the proof of Theorem 1 randomization does not give us back RSD. This in contrast to randomization of deterministic mechanisms satisfying ex-post efficiency, strategy-proofness and non-bossiness which gives us back RSD (Bade, 2020). Furthermore, as bounded invariance is preserved for weighted versions for hierarchical exchange rules, Theorem 2 shows that any such weighted version of hierarchical exchange rules is not sd-dominated by another strategy-proof and boundedly invariant mechanism.

### 5.2 Deferred Acceptance

In certain applications objects are allocated according agents' priorities – a priority order for an object o is a linear order  $\succeq_o$  over N, and a priority structure is given by  $\succeq = (\succeq_o)_{o \in O}$ .

For priority-based object allocation, the two well known and studied mechanisms are TTC and Deferred Acceptance (DA). TTC is the subclass of inheritance rules for which inheritance depends only on  $\succeq$ , i.e., for any  $\sigma$  and  $o \in \overline{H}_{\sigma}$ , the  $\succeq_{o}$ -highest ranked agent in  $\overline{I}_{\sigma}$  owns object o. TTC is efficient and strategy-proof (but unstable) whereas DA is stable and strategy-proof (but inefficient).<sup>25</sup> This is the well-known stability-efficiency tradeoff in deterministic assignment problems and depending on the application under consideration, either TTC or DA is recommended, which was studied in detail for school choice by Abdulkadiroğlu and Sönmez (2003b).

#### Deferred Acceptance Algorithm.

Let  $\succeq$  be a priority structure and R be a preference profile.

Step 1. Let each agent propose to his most preferred object. Each object tentatively accepts (if any) the highest priority-ranked agent among all proposals and rejects the rest. Stop if no agent is rejected (and all (tentative) acceptances become the outcome), and otherwise continue with Step 2.

Step r. Each rejected agent proposes to the next most preferred object (to which he has not proposed yet). Each object tentatively accepts (if any) the highest priority-ranked agent among all proposals and tentative acceptance, and rejects the rest. Stop if no agent is rejected (and all (tentative) acceptances become the outcome), and otherwise continue with Step r + 1.

bossiness. This is not the case which we show in the Appendix using an example from Svensson (1999, p.563).

<sup>&</sup>lt;sup>25</sup>An assignment  $\mu$  is stable for R and  $\succeq$  if there does not exist an agent-object pair (i, x) such that  $xP_i\mu_i$ and either x is not assigned to any agent or for  $\mu_j = x$  we have  $i \succ_x j$ , i.e. i has higher priority than j to be assigned x.

**Definition 4** A priority structure  $\succeq$  contains a cycle if there exist distinct  $a, b \in O$  and distinct  $i, j, k \in N$  such that  $i \succ_a j \succ_a k$  and  $k \succ_b i$ . The priority structure is acyclic if it does not contain any cycle.

Ergin (2002) has shown that acyclicity characterizes the priority structures for which DA is efficient meaning that the efficiency-stability tradeoff disappears for DA. Below we show that equivalently bounded invariance is crucial for DA to be efficient.<sup>26</sup>

**Theorem 4** Let  $\succeq$  be a priority structure and consider the associated DA mechanism. The following are equivalent:

- (i) DA satisfies bounded invariance.
- (ii) DA satisfies efficiency.
- (iii) DA satisfies non-bossiness.
- $(iv) \succcurlyeq is acyclic.$

**Proof.** (i) $\Rightarrow$ (iv): Suppose that  $\succeq$  contains a cycle, say  $1 \succ_a 2 \succ_a 3$  and  $3 \succ_b 1$ . Let  $O = \{a, b, c, o_1, \ldots, o_{n-3}\}$   $R_1 : bac \ldots$ ,  $R_2 : acb \ldots$ , and  $R_3 : abc \ldots$  and  $R_i : o_{i-3} \ldots$  for  $i \neq 1, 2, 3$ . We have (where  $i \neq 1, 2, 3$ )  $DA(R) = \begin{pmatrix} 1 & 2 & 3 & i \\ a & c & b & o_{i-3} \end{pmatrix}$  and DA is inefficient (as agents 1 and 3 would prefer to swap their assigned objects). Now let  $R'_3 : acb \ldots$  and we obtain (where  $i \neq 1, 2, 3$ )  $DA(R'_3, R_{-3}) = \begin{pmatrix} 1 & 2 & 3 & i \\ b & a & c & o_{i-3} \end{pmatrix}$ , a violation to bounded invariance as  $DA_1(R) = a = DA_2(R'_3, R_{-3})$  and both  $R_3$  and  $R'_3$  rank a at the top. Thus,  $\succeq$  must be acyclic.

(ii) $\Leftrightarrow$ (iii) $\Leftrightarrow$ (iv): This follows from the Ergin (2002, Theorem).

 $(iv) \Rightarrow (i)$ : If  $\succeq$  is acyclic, then DA becomes a hierarchical exchange rule where inheritance of objects follows the priority structure  $\succeq$ . Thus, by Theorem 3, DA satisfies bounded invariance.

# 6 Tie-breaking in Top-Trading-Cycles

Although the priority orders considered above were strict, in applications such as school choice they are often weak, i.e.,  $\succeq_o$  may contain ties when agents have equal priority for

<sup>&</sup>lt;sup>26</sup>Ergin (2002) considers a general assignment model where both objects come with capacities and agents have the option to be unassigned (which may be preferred to certain objects). Then a "scarcity" condition has to be added to Definition 4. One can check that Theorem 4 remains unchanged in this more general setup.

object *o*. This raises the question of how to break ties, which has been discussed in detail for deterministic mechanisms given the desirability of stability, strategy-proofness and constrained efficiency among stable assignments<sup>27</sup> (where for strict priorities DA satisfies all these properties). For instance, Abdulkadiroğlu et al. (2009) show that DA with tie-breaking is not Pareto-dominated by any strategy-proof deterministic mechanism while Erdil and Ergin (2008) show the non-existence of stable, strategy-proof and constrained efficient deterministic mechanisms. Ehlers and Erdil (2010) characterize the priority structures for which constrained efficiency among stable matchings is equivalent to efficiency while Ehlers and Westkamp (2018) characterize the weak priority structures for which strategy-proofness, stability and constrained efficiency are compatible for deterministic mechanisms. Han (2018) characterizes priority structures for which strategy-proofness, stability and constrained efficiency are compatible for deterministic mechanisms. Han (2018) characterizes priority structures for which strategy-proofness, stability and efficiency are compatible for deterministic mechanisms. Han (2018) characterizes priority structures for which strategy-proofness, stability and efficiency are compatible for deterministic mechanisms.

There is continued discussion whether to use single-tie-breaking (STB) or multiple-tiebreaking (MTB) for weak priority structures, see Pathak (2017) for a detailed account. STB means that ties are broken in the same way for all objects' priority orderings while MTB means that ties are broken in possible different ways for objects' priority orderings. For deterministic mechanisms, there is no clear recommendation for using STB versus MTB for DA – while some agents may be better off at some preference profile, others will be worse off, possibly at another profile. Recent contributions have considered the ex-ante welfare effects of STB and MTB before tie-breaking, i.e., for random mechanisms where for STB any tie-breaking order is chosen with equal probability whereas for MTB any profile of tiebreaking orders is chosen with equal probability. For given preference profiles, but before ties are broken, Han (2020) characterized the priority-structures for which STB-DA is expost constrained efficient. For an ex-ante analysis – before agents' preference realization and tie-breaking - in large markets with random preferences, such as Ashlagi et al. (2019), Ashlagi and Nikzad (2020), Arnosti (2023), and Allman et al. (2023) provide a detailed rank distribution comparison analysis (to which choices agents are matched) for STB-DA and MTB-DA, and identify circumstances under which STB-DA is superior to MTB-DA.

Most importantly, this has not been discussed in detail for the TTC-mechanism with tie-breaking. Obviously, as TTC-mechanisms are efficient, unlike DA the issue of Paretodominance does not arise among deterministic mechanisms. Once ties are broken randomly like in STB or MTB this is no longer true. For instance, for single tie-breaking (STB) we take an order of agents, denoted by > to break ties identically: for all  $o \in O$ , if  $i \sim_o j$ and  $i \geq_o j$ , then  $i \succ_o^> j$ . For the obtained strict priority structure we apply TTC. Now choosing any tie-breaking order with equal probability 1/n!, we obtain STB-TTC which is

<sup>&</sup>lt;sup>27</sup>I.e., assignments which are not Pareto-dominated by any other stable assignment.

ex-post efficient, strategy-proof and boundedly invariant (by Theorem 3), and hence by our second main result – Theorem 2 – not stochastically dominated by any other strategy-proof and boundedly invariant mechanism. The same is true for multiple tie-breaking whereby we choose different tie-breaking orders for different objects, i.e., for any object o we have a tiebreaking order  $\geq_o$  to break ties in  $\succeq_o$ , and now any profile of tie-breaking orders (one for each object) ( $\geq_o$ )<sub> $o \in O$ </sub> we apply TTC for the obtained strict priority structure. Now MTB-TTC is obtained by choosing any profile of tie-breaking orders with equal probability  $1/(n!)^n$ . Again by Theorem 2 MTB-TTC is not stochastically dominated by any other strategy-proof and boundedly invariant mechanism.

#### **Corollary 1** Let $\succeq$ be a weak priority structure.

- (i) No boundedly invariant and strategy-proof mechanism sd-dominates STB-TTC.
- (ii) No boundedly invariant and strategy-proof mechanism sd-dominates MTB-TTC.

Furthermore, this remains true when different tie-breaking orders are chosen with different weights, i.e., again such a weighted MTB-TTC is not stochastically dominated by any other strategy-proof and boundedly invariant mechanism. It follows from Theorem 2 that weighted versions of MTB-TTC (including weighted versions of STB-TTC) do not sddominate each other as they all satisfy ex-post efficiency, strategy-proofness and bounded invariance. For random versions of STB-TTC and MTB-TTC, Corollary 1 is parallel to Abdulkadiroğlu et al. (2009) for the deterministic versions of STB-DA and MTB-DA (but as explained above, they do not hold any longer for random versions of STB-DA and MTB-DA).

Note that when all agents have equal priority at all objects, then STB-TTC coincides with RSD, i.e., as a special case we obtain again that RSD is not sd-dominated in the class of boundedly invariant strategy-proof mechanisms.

## 7 Conclusion

Instead of reporting ordinal preferences, one might ask agents to report cardinal utility functions, assuming that they evaluate random assignments according to their expected utilities. We implicitly assume ordinality of mechanisms, i.e., constrain random assignments to be the same across cardinal utility profiles which induce identical ordinal preferences. For applications ordinality is a natural requirement as it facilitates reporting, when agents are unable to determine their exact utilities but are able to compare individual objects. Of course, allowing cardinal reports but imposing ordinality yields the same result as imposing ordinal preference reports. In particular, in such contexts RSD is not dominated by any mechanism satisfying ordinality, strategy-proofness and bounded invariance. This is positive answer on the acceptable domain and addresses a question raised by Zhou (1990), who showed that in the cardinal framework no mechanism satisfies equal treatment of equals, strategy-proofness and ex-ante efficiency. The latter postulates always to choose a random assignment which is not Pareto dominated in terms of expected utility by any other. It is clear that in the cardinal context the properties of ordinality, equal treatment of equals and ex-ante efficiency are incompatible: as a simple example, let  $N = \{1, 2, 3\}$ ,  $O = \{a, b, c\}$ ,  $u_1 = (u_{1a}, u_{1b}, u_{1c}) = (1, 1 - \epsilon, 0) = u_2$  and  $u_3 = (1, \epsilon, 0)$  where  $\epsilon > 0$  is small; when all agents have utility function  $u_3$  equal treatment of equals requires each agent to obtain a with probability one third, and similarly, when all agents have utility function  $u_1$  equal treatment of equals requires each agent to obtain c with probability one third; now ordinality requires for the profile  $(u_1, u_2, u_3)$  that each agent obtains any object with probability  $\frac{1}{3}$ , which is dominated in terms of expected utility by assigning agent 3 object b with probability one, and assigning agents 1 and 2 objects a and c each with probability one half.

The last example shows the disrelation of Zhou's result and the impossibility results in the ordinal framework with respect to efficiency, equity and strategy-proofness. Ordinality, sd-efficiency and envy-freeness are compatible as PS satisfies all those properties. As soon as strategy-proofness is added, we obtain an incompatibility, which is robust when weakening sd-efficiency to ex-post efficiency, or envy-freeness to equal treatment of equals.

Theorem 1 invalidated the conjecture that RSD was characterized by ex-post efficiency, equal treatment of equals and strategy-proofness.<sup>28</sup> Pycia and Troyan (2021) recently showed that RSD is characterized by symmetry, efficiency, and obvious strategyproofness among all mechanisms that, roughly speaking, can be represented as a symmetrization of an extensive-form game where in each stage, one agent is allowed to pick one house from a subset of the remaining houses or "pass" on this opportunity. For the assignment of one object, Ehlers (2002) characterized the uniform random dictatorship mechanism by ex-post efficiency, envy-freeness and strategy-proofness.

Another strand of the literature studies large markets. In particular one may enlarge markets in two different ways: either by keeping the set of object types fixed and adding copies to match an increasing number of agents, or by considering a sequence of economies where the number of distinct agents and the number of distinct objects grow at the same rate. First, when we add object copies, Liu and Pycia (2016, Theorem 2) have shown that any two symmetric and "regular"<sup>29</sup> mechanisms, which are asymptotically strategy-proof

 $<sup>^{28}\</sup>mathrm{Brandt}$  et al. (2023) claimed to have found an affirmative answer to this question for five agents via computational methods.

<sup>&</sup>lt;sup>29</sup>Loosely speaking, this means that agents cannot change to "too much" the random assignments of other agents (in terms of probability shares) as the market becomes large.

and asymptotically efficient, coincide asymptotically, i.e., they choose the same allocations in the limit. For instance, this implies asymptotic coincidence of RSD<sup>30</sup> and PS (which was first shown by Che and Kojima (2010)), and that RSD and, respectively, PS satisfy ex-post efficiency and asymptotically both strategy-proofness and envy-freeness. In some sense, then, it does not matter in the large whether we choose RSD or PS (or any other mechanism satisfying the above three properties). However when we consider economies with a large number of distinct agents and distinct objects,<sup>31</sup> Manea (2009) has shown that RSD is sd-efficient with probability approaching zero, and hence RSD and PS diverge with probability one. Thus, continued discussions in real-life markets show the importance of the choice of the random assignment mechanism to be implemented. As we have shown, RSD cannot be improved in an unambiguous way while maintaining our two basic properties.

# References

- Atila Abdulkadiroğlu and Tayfun Sönmez. Ordinal efficiency and dominated sets of assignments. *Journal of Economic Theory*, 112(1):157–172, 2003a.
- Atila Abdulkadiroğlu and Tayfun Sönmez. School choice: A mechanism design approach. American Economic Review, 93(3):729–747, 2003b.
- Atila Abdulkadiroğlu, Parag A Pathak, and Alvin E Roth. Strategy-proofness versus efficiency in matching with indifferences: Redesigning the NYC high school match. American Economic Review, 99(5):1954–1978, 2009.
- Maxwell Allman, Itai Ashlagi, and Afshin Nikzad. On rank dominance of tie-breaking rules. *Theoretical Economics*, 18(2):707–748, 2023.
- Samson Alva, Eun Jeong Heo, and Vikram Manjunath. Efficiency in random allocation with ordinal rules, 2024.
- Nick Arnosti. Lottery design for school choice. Management Science, 69(1):244–259, 2023.
- Itai Ashlagi and Afshin Nikzad. What matters in school choice tie-breaking? how competition guides design. *Journal of Economic Theory*, 190:105120, 2020.

 $<sup>^{30}</sup>$ RSD is regular, provided the number of copies for each object type grows at the same rate as the number of agents, e.g., in replica economies.

 $<sup>^{31}</sup>$ E.g., in a school choice context, this would describe a scenario where the number of applicants and schools grows, but the capacity of individual schools is bounded.

- Itai Ashlagi, Afshin Nikzad, and Assaf Romm. Assigning more students to their top choices: A comparison of tie-breaking rules. *Games and Economic Behavior*, 115:167–187, 2019.
- Sophie Bade. Random serial dictatorship: the one and only. *Mathematics of Operations* Research, 45(1):353–368, 2020.
- Christian Basteck and Lars Ehlers. Strategy-proof and envyfree random assignment. *Journal* of *Economic Theory*, 209:105618, 2023.
- Garrett Birkhoff. Three observations on linear algebra. Univ. Nac. Tacuman, Rev. Ser. A, 5:147–151, 1946.
- Anna Bogomolnaia. Random assignment: redefining the serial rule. *Journal of Economic Theory*, 158:308–318, 2015.
- Anna Bogomolnaia and Eun Jeong Heo. Probabilistic assignment of objects: Characterizing the serial rule. *Journal of Economic Theory*, 147:2072–2082, 2012.
- Anna Bogomolnaia and Hervé Moulin. A new solution to the random assignment problem. Journal of Economic Theory, 100:295–328, 2001.
- Felix Brandt, Matthias Greger, and René Romen. Characterizing random serial dictatorship. Working Paper March 21, 2023.
- Yeon-Koo Che and Fuhito Kojima. Asymptotic equivalence of probabilistic serial and random priority mechanisms. *Econometrica*, 78(5):1625–1672, 2010.
- Lars Ehlers. Probabilistic allocation rules and single-dipped preferences. Social Choice and Welfare, 19:325–348, 2002.
- Lars Ehlers and Aytek Erdil. Efficient assignment respecting priorities. *Journal of Economic Theory*, 145(3):1269–1282, 2010.
- Lars Ehlers and Alexander Westkamp. Strategy-proof tie-breaking in matching with priorities. *Theoretical Economics*, 13(3):1009–1041, 2018.
- Aytek Erdil. Strategy-proof stochastic assignment. *Journal of Economic Theory*, 151:146–162, 2014.
- Aytek Erdil and Haluk Ergin. What's the matter with tie-breaking? improving efficiency in school choice. *American Economic Review*, 98(3):669–689, 2008.

- Haluk I Ergin. Efficient resource allocation on the basis of priorities. *Econometrica*, 70(6): 2489–2497, 2002.
- Allan Gibbard. Manipulation of schemes that mix voting with chance. *Econometrica*, 45(3): 665–681, 1977.
- Xiang Han. Stable and efficient resource allocation under weak priorities. *Games and Economic Behavior*, 107:1–20, 2018.
- Xiang Han. On the efficiency and fairness of deferred acceptance with single tie-breaking. Available at SSRN 3516424, 2020.
- Tadashi Hashimoto, Daisuke Hirata, Onur Kesten, Morimitsu Kurino, and M. Utku Unver. Two axiomatic approaches to the probabilistic serial mechanism. *Theoretical Economics*, 9:253–277, 2014.
- Akshay-Kumar Katta and Jay Sethuraman. A solution to the random assignment problem on the full preference domain. *Journal of Economic Theory*, 131:231–250, 2006.
- Qingmin Liu and Marek Pycia. Ordinal efficiency, fairness, and incentives in large markets. Working Paper, 2016.
- Mihai Manea. Asymptotic ordinal inefficiency of random serial dictatorship. *Theoretical Economics*, 4:165–197, 2009.
- Andrew McLennan. Ordinal efficiency and the polyhedral separating hyperplane theorem. Journal of Economic Theory, 105:435–449, 2002.
- Alexander S Nesterov. Fairness and efficiency in strategy-proof object allocation mechanisms. Journal of Economic Theory, 170:145–168, 2017.
- Szilvia Pápai. Strategyproof assignment by hierarchical exchange. *Econometrica*, 68(6): 1403–1433, 2000.
- Parag A Pathak. What really matters in designing school choice mechanisms. Advances in Economics and Econometrics, 1(12):176–214, 2017.
- Marek Pycia and Peter Troyan. A theory of simplicity in games and mechanism design. Working paper series/Department of Economics, (393), 2021.
- Marek Pycia and Peter Troyan. Strategy-proof, efficient, and fair allocation: Beyond random priority. *Working Paper*, 2023.

- Marek Pycia and M Utku Ünver. Incentive compatible allocation and exchange of discrete resources. *Theoretical Economics*, 12(1):287–329, 2017.
- Priyanka Shende and Manish Purohit. Strategy-proof and envy-free mechanisms for house allocation. *Journal of Economic Theory*, 213:105712, 2023.
- Lars-Gunnar Svensson. Strategy-proof allocation of indivisible goods. Social Choice and Welfare, 16:557–567, 1999.
- Jun Zhang. Efficient and fair assignment mechanisms are strongly group manipulable. *Jour*nal of Economic Theory, 180:167–177, 2019.
- Lin Zhou. On a conjecture by Gale about one-sided matching problems. *Journal of Economic Theory*, 52:123–135, 1990.

#### APPENDIX.

# A General Version of Theorem 1

Below we allow for the possibility of unequal numbers of agents and objects. With fewer objects than agents, some agents may remain unassigned. All our definitions then extend in a straightforward way. Theorem 1 is a corollary from this more general result.

We further strengthen symmetry to anonymity where agents' names are treated equally: a mechanism f satisfies anonymity if for any permutation  $\pi : N \to N$  of agents and for any profile R, we have  $f_i(R) = f_{\pi(i)}((R_{\pi(i)})_{i \in N})$ .

In addition, the constructed mechanism is immune to renaming objects: a mechanism f satisfies neutrality if for any permutation  $\tau : O \to O$  of objects and for any profile R, we define  $\tau(R) = (\tau(R_i)_{i \in N}) \in \mathbb{R}^N$  such that for  $o, o' \in O$ ,  $\tau(o)\tau(R_i)\tau(o')$  iff  $oR_io'$ , and we have for all  $i \in N$  and all  $o \in O$ ,  $f_{io}(R) = f_{i\tau(o)}(\tau(R))$ .

**Theorem 5** For  $|N| \ge 5$  and  $|O| \ge 5$ , there exist mechanisms satisfying ex-post efficiency, anonymity, neutrality and strategy-proofness, which are not welfare equivalent to random serial dictatorship.

**Proof.** We begin with five agents and five objects, i.e., let  $N = \{1, \ldots, 5\}$  and  $O = \{a, b, c, d, e\}$ .

First, we define the following mechanism  $g^{5}$  whereby agents 1, 2, 3 and 4 are ranked arbitrarily and choose in that order (as in RSD for four agents) and afterwards agent 5 receives

the remaining object. Making it symmetric for agents (by choosing  $g^{\underline{i}}$  with probability  $\frac{1}{5}$ ) we get back RSD (as then any order is chosen with equal probability  $\frac{1}{5!}$ ), i.e.,  $RSD = \frac{1}{5} \sum_{i \in N} g^{\underline{i}}$ .

Second, let  $R_2, R_3, R_4$  be as follows

$R_2$	$R_3$	$R_4$
С	С	С
a	b	e
d	d	d
e	e	a
b	a	b

It will turn out to be crucial that the same object d is ranked below c and a for agent 2 and below c and b for agent 3, and that agent 4 ranks the different object e below c. We define the mechanism  $g^{1-5}$  whereby agent 1 is improved over  $g^5$ , agents 2, 3 and 4 receive identical random assignments under  $g^{1-5}$  and  $g^5$ , and agent 5 is worse off or better off under  $g^{1-5}$ compared to  $g^5$ .

We decompose the preference domain for agent 1 as the disjoint union of the following three sets:

$$\mathcal{R}_{a}^{1} = \{R_{1} : aP_{1}bP_{1}x \text{ for all } x \in O \setminus \{a, b, c\}\}$$
  

$$\mathcal{R}_{b}^{1} = \{R_{1} : bP_{1}aP_{1}x \text{ for all } x \in O \setminus \{a, b, c\}\}$$
  

$$\hat{\mathcal{R}}^{1} = \{R_{1} : \text{ there exists } x \in O \setminus \{a, b, c\} \text{ such that } xP_{1}a \text{ or } xP_{1}b\}.$$

For all  $Q \in \mathcal{R}^N$ , let

$$g_i^{1-\underline{5}}(Q) = g_{\underline{i}}^{\underline{5}}(Q) \text{ for } i = 2, 3, 4.$$
 (1)

Moreover, for any profile Q, if  $Q_{-15} \neq (R_2, R_3, R_4)$  or  $Q_1 \in \hat{\mathcal{R}}^1$ , then  $g^{1-\underline{5}}(Q) = g^{\underline{5}}(Q)$ .

Otherwise, suppose  $Q_1 \in \mathcal{R}^1_a \cup \mathcal{R}^1_b$  and  $Q_{-15} = (R_2, R_3, R_4)$ . Note that once we have defined  $g_1^{1-\underline{5}}(Q)$ , then  $g_5^{1-\underline{5}}(Q)$  is the residual given (1).

If  $Q_1 \in \mathcal{R}^1_a$ , then under  $g^5(Q)$  agent 1 receives their most preferred object from  $O \setminus \{a, b, c\}$  for the orders 4-2-3-1-5 and 4-3-2-1-5 (where any such order is chosen with probability 1/24), i.e., with probability 1/12. Similarly, object b is assigned to agent 5 for the orders 3 - 1 - 4 - 2 - 5 and 3 - 1 - 2 - 4 - 5, i.e., at least with probability 1/12.

Then let  $g_1^{1-5}(Q) = g_1^5(Q) + 1/12b - 1/12x$  where x is 1's most preferred object from  $O \setminus \{a, b, c\}$ . Note that then  $g_1^{1-5}(Q)$  strictly sd-improves agent 1 over  $g_1^5(Q)$  (and if  $Q_5 : be \ldots$ , then  $g_1^5(Q)$  strictly sd-dominates  $g_1^{1-5}(Q)$ , i.e., agent 5 is unambiguously worse off, and if  $Q_5 : \ldots b$ , then  $g_1^{1-5}(Q)$  strictly sd-dominates  $g_1^5(Q)$ , i.e., agent 5 is unambiguously better

off).

Analogously, if  $Q_1 \in \mathcal{R}_b^1$ , then let  $g_1^{1-\underline{5}}(Q) = g_1^{\underline{5}}(Q) + 1/12a - 1/12x$  where x is 1's most preferred object from  $O \setminus \{a, b, c\}$ .

It remains to show that  $g^{1-5}$  is strategy-proof and ex-post efficient. For strategy-proofness, it is obvious that agents 2, 3, 4 and 5 cannot gain from manipulation. Let  $Q \in \mathcal{R}^N$ . If  $Q_{-15} \neq (R_2, R_3, R_4)$ , then agent 1 cannot gain from manipulation as  $g_1^{1-5}(Q) = g_1^5(Q)$  and  $g_1^{1-5}(R_1, Q_{-1}) = g_1^5(R_1, Q_{-1})$  for all  $R_1$ . Thus, let  $Q_{-15} = (R_2, R_3, R_4)$ . By Lemma 2 of Gibbard (1977) it suffices to consider pairwise switches of objects ranked adjacently, i.e., compare the random assignment of agent 1 when reporting  $Q_1$  and when reporting  $R_1 = Q_1^{y\leftrightarrow z}$  for two objects  $y, z \in O$  ranked adjacent in  $Q_1$ .<sup>32</sup>

Suppose that  $Q_1 \in \mathcal{R}_a^1$ . Note that by ex-post efficiency and as agent 5 chooses last, agent 1 receives object c with probability zero when c is not ranked first in  $Q_1$ . Suppose c is not ranked first in  $Q_1$ . Then by construction  $g_1^{1-5}(Q) = (18a + 6b)/24$ . If  $R_1 \in \mathcal{R}_a^1 \cup \mathcal{R}_b^1$ , yet cis not ranked first under  $R_1$ , then 1's random assignment either unchanged when reporting  $R_1$  instead of  $Q_1$  or changed to (18b + 6a)/24. If instead  $R_1 \in \mathcal{R}_a^1 \cup \mathcal{R}_b^1$ , and c is now ranked first under  $R_1$ , then  $g_1^{1-5}(R_1, Q_{-1}) = (6c + 12a + 6b)/24$ . The case where c is ranked first under  $Q_1$  and  $R_1 \in \mathcal{R}_a^1 \cup \mathcal{R}_b^1$  is analysed analogously.

Suppose  $R_1 \in \hat{\mathcal{R}}^1$ . Then  $g_1^{1-\underline{5}}(R_1, Q_{-1}) = g_{\underline{1}}^{\underline{5}}(R_1, Q_{-1})$ . But then  $g_1^{1-\underline{5}}(Q)$  by construction stochastically  $Q_1$ -dominates  $g_{\underline{1}}^{\underline{5}}(Q)$ , which, by strategy-proofness of  $g_{\underline{5}}$ , stochastically  $Q_1$ -dominates  $g_{\underline{1}}^{\underline{5}}(R_1, Q_{-1}) = g_1^{1-\underline{5}}(R_1, Q_{-1})$ . Thus, reporting  $R_1$  leads to a  $Q_1$ -dominated random assignment.

This completes the analysis for  $Q_1 \in \mathcal{R}^1_a$ . The case  $Q_1 \in \mathcal{R}^1_b$  is analysed analogously.

For  $Q_1 \in \hat{\mathcal{R}}^1$ , if  $R_1 \in \hat{\mathcal{R}}^1$  then  $g_1^{1-5}$  and  $g_1^5$  coincide both at Q and at  $(R_1, Q_{-1})$  – hence strategy-proofness of  $g_1^5$  implies that  $g_1^{1-5}(Q)$  stochastically  $Q_1$ -dominates  $g_1^5(R_1, Q_{-1})$ . If  $R_1 \in \mathcal{R}_a^1$ , then the pairwise swap of two objects must have involved b (ranked below a under both  $Q_1$  and  $R_1$ ) and object  $x \in O \setminus \{a, b, c\}$  such that  $xQ_1d$  and  $xQ_1e$ , i.e.,  $aQ_1xQ_1b$  and  $aR_1bR_1x$ . By construction, when reporting  $R_1$  instead of  $Q_1$ , this only moves probability mass from x to b – hence  $g_1^{1-5}(Q)$  stochastically  $Q_1$ -dominates  $g_1^5(R_1, Q_{-1})$ . Again, the case  $R_1 \in \mathcal{R}_b^1$  is analysed analogously.

For completeness below we just list the first three objects in  $R_1$  (where  $x \in \{d, e\}$  is such that  $xR_1d$  and  $xR_1e$ ),  $R_1 = Q_1^{y\leftrightarrow z}$  (where possibly  $x \in \{z, y\}$ ), and  $Q_1 \in \mathcal{R}_a^1 \cup \mathcal{R}_b^1$ . This implies that x cannot be ranked at the top of  $R_1$ , and x cannot be ranked second while c is ranked third in  $R_1$ ):

If  $aP_1bP_1x$  or  $aP_1cP_1b$ , then her assignment is (18a + 6b)/24.

<sup>&</sup>lt;sup>32</sup>If  $R_1 : uvwyzx$ , then  $Q_1 : uvwzyx$ .

If  $bP_1aP_1x$  or  $bP_1cP_1a$ , then her assignment is (18b + 6a)/24. If  $cP_1aP_1b$ , then her assignment is (6c + 12a + 6b)/24. If  $cP_1bP_1a$ , then her assignment is (6c + 12b + 6a)/24. If  $aP_1xP_1b$ , then her assignment is (18a + 4x + 2b)/24. If  $aP_1cP_1x$ , then her assignment is (18a + 6x)/24. If  $bP_1cP_1x$ , then her assignment is (18b + 6x)/24. If  $bP_1xP_1a$ , then her assignment is (18b + 4x + 2a)/24. If  $cP_1aP_1x$ , then her assignment is (6c + 12a + 6x)/24. If  $cP_1aP_1x$ , then her assignment is (6c + 12a + 6x)/24. If  $cP_1bP_1x$ , then her assignment is (6c + 12b + 6x)/24.

Under the first four announcements agent 1 receives objects d and e with probability zero, and at each announcement agent 1 receives with probability one his first three objects. A straightforward pairwise comparison of these ten outcomes verifies that at each preference ranking  $Q_1 \in \mathcal{R}^1_a \cup \mathcal{R}^1_b \cup \hat{\mathcal{R}}^1$ , truthful revelation (weakly or strongly) first-order stochastically dominates untruthful revelation.

Finally, we verify ex-post efficiency of  $g^{1-5}$ . Here it is crucial that the same object d is ranked below c and a for agent 2 and below c and b for agent 3, and that agent 4 ranks the different object e below c. Improving 1's assignment when  $R_1 \in \mathcal{R}^1_a$  involves increasing her share of b by 1/12, while holding unchanged the assignment of agents 2, 3, and 4. This can be done by replacing the assignments on the left below with the assignments on the right with probability share 1/12 each. Each assignment on the left is realized with probability 1/12 or more, therefore this improvement is indeed feasible. And the assignments on the right are efficient, therefore ex-post efficiency is preserved (where  $\{d, e\} = \{x, y\}$  and  $xP_1y$ )

$$\begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ x & a & b & c & y \end{pmatrix} + \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ a & d & c & e & b \end{pmatrix} \rightarrow \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ b & a & c & e & d \end{pmatrix} + \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ a & d & b & c & e \end{pmatrix}$$

Note that the first assignment is obtained for the orders 4-2-3-1-5 and 4-3-2-1-5, the second one for the orders 3-1-2-4-5 and 3-1-4-2-5, the third one for the orders 3-2-1-4-5 and 3-2-4-1-5, and the fourth one for the orders 4-1-2-3-5 and 4-1-3-2-5.

The argument for the case when  $R_1 \in \mathcal{R}_b^1$  is analogous but for completeness we verify it below. Improving 1's assignment when  $R_1 \in \mathcal{R}_b^1$  involves increasing her share of *a* by 1/12, while holding unchanged the assignment of agents 2, 3, and 4. This can be done by replacing the assignments on the left below with the assignments on the right with probability share 1/12 each. Each assignment on the left is realized with probability 1/12 or more, therefore this improvement is indeed feasible. And the assignments on the right are efficient, therefore ex-post efficiency is preserved (where  $\{d, e\} = \{x, y\}$  and  $xP_1y$ )

$$\begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ x & a & b & c & y \end{pmatrix} + \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ b & c & d & e & a \end{pmatrix} \rightarrow \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ a & c & b & e & d \end{pmatrix} + \begin{pmatrix} 1 & 2 & 3 & 4 & 5 \\ b & a & d & c & e \end{pmatrix}$$

The first assignment is obtained for the orders 4 - 2 - 3 - 1 - 5 and 4 - 3 - 2 - 1 - 5, the second one for the orders 2 - 1 - 3 - 4 - 5 and 2 - 1 - 4 - 3 - 5, the third one for the orders 2 - 3 - 1 - 4 - 5 and 2 - 3 - 4 - 1 - 5, and the fourth one for the orders 4 - 1 - 2 - 3 - 5 and 4 - 1 - 3 - 2 - 5.

Third, the mechanism  $g^{1-5}$  treats agent 1 differently in comparison to agents 2, 3 and 4. In order to recover equal treatment of equals among agents 1, 2, 3, and 4, we appeal to randomization again.

Let  $\pi$  be a permutation of the agents with agent 5 staying put, i.e., let  $\pi : N \to N$ be a bijection such that  $\pi(5) = 5$ . Then  $\pi(g^{1-5})$  is defined via changing the roles of the agents in mechanism  $g^{1-5}$  according to the permutation  $\pi$ . Denoting with  $\Pi^5$  the set of all permutations of N where agent 5 stays put, we define  $h^{5}$  as

$$h^{\underline{5}} = \frac{1}{4!} \sum_{\pi \in \Pi^5} \pi(g^{1-\underline{5}}).$$

Then  $h^{\underline{5}}$  inherits ex-post efficiency and strategy-proofness from  $g^{1-\underline{5}}$ , and agents 1, 2, 3, and 4 are treated symmetrically. Note also that  $\sum_{\pi \in \Pi^5} \pi(g^{\underline{5}}) = g^{\underline{5}}$ , and hence agents 1, 2, 3 and 4 are better off under  $h^{\underline{5}}$  and agent 5 is worse off or better off under  $h^{\underline{5}}$  (when compared to  $g^{\underline{5}}$ ).

Fourth, in order to recover anonymity (and, respectively, symmetry) completely, let  $h = \frac{1}{5} \sum_{i \in N} h^{\underline{i}}$ . Again, h inherits ex-post efficiency and strategy-proofness from  $h^{\underline{i}}$ , and satisfies anonymity and symmetry.

Fifth, we show that h does not coincide with RSD, i.e., there exists a profile Q such that  $h_i(Q) \neq RSD_i(Q)$  for some  $i \in N$ . Thus, there exist other random mechanisms (in terms of probability shares) satisfying ex-post efficiency, symmetry and strategy-proofness. Let  $Q \in \mathcal{R}^N$  be such that  $Q_1$ : *abecd*,  $Q_{-15} = (R_2, R_3, R_4)$  and  $Q_5$ : *edcab*.

Then  $g_1^{1-5}(Q) = (18a+6b)/24 \neq (18a+4b+2e)/24 = g_1^5(Q)$ . Now let us consider  $g^{i-\underline{j}}(Q)$ and  $g^{\underline{j}}(Q)$ , i.e., where we permute the roles of agents. For i = 5 and j = 1 the two coincide, since agent 5 does not rank a and b over d and e. Finally, if  $\{i, j\} \neq \{1, 5\}$  then again  $g^{i-\underline{j}}(Q) = g^{\underline{j}}(Q)$  since either 1 or 5 is now in the role of agents 2, 3, or 4 in  $g^{1-\underline{5}}$  but neither 1 nor 5 ranks c first. But now it follows that  $h_{1e}(Q) \neq RSD_{1e}(Q)$ , and h(Q) stochastically Q-dominates RSD(Q) (as agents 1 and 5 are better off and agents 2, 3 and 4 receive identical random assignments).

Finally, we show that neutrality can be recovered from the above mechanism. Let  $\tau$ :  $O \to O$  be a renaming of the objects and denote by  $\Gamma$  the set of all such bijections. Let  $\tau(h)$ denote the permuted mechanism where the names of the objects in h are changed according to  $\tau$ , and let  $\overline{h} = \frac{1}{5!} \sum_{\tau \in \Gamma} \tau(h)$ . But then  $\overline{h}$  inherits all the properties from h and satisfies neutrality. Furthermore, for the above profile, we continue to have  $\overline{h}_{1e}(Q) \neq RSD_{1e}(Q)$ .

Lastly, suppose that there at least five agents and at least five objects, i.e.,  $N = \{1, \ldots, n\}$ with  $n \geq 5$  and  $O = \{a, b, c, d, e, o_1, \ldots, o_{|O|-5}\}$  with  $|O| \geq 5$  where possibly  $|N| \neq |O|$ . We then define the mechanism  $g^{56\cdots n}$  by letting choose agents 1 - 4 in a random order and then the remaining agents in the order  $5 - 6 - \cdots - n$ . Again permuting gives us back RSD. For the mechanism  $g^{1-\underline{56}\cdots n}$  we add to the preferences  $R_2$ ,  $R_3$  and  $R_4$  the other objects in the same order  $o_1 - \cdots - o_{|O|-5}$  at the bottom. Now for agent 1 an improvement is applied under the same conditions (where no  $x \in O \setminus \{a, b, c\}$  shall be ranked above a or b). One can check again strategy-proofness and ex-post efficiency, and make the mechanism symmetric. In showing that the new mechanism does not coincide with RSD, let the preferences of agents 1-5 be as above in profile Q with the other objects being ranked in the same order  $o_1 - \cdots - o_{|O|-5}$  at the bottom and let agent i (with  $i \geq 5$ ) have the same preference as agent 5. Note that then the above improvement is applied for agent 1 in the mechanism  $g^{\underline{56}\cdots n}$  but when at least one agent i with  $i \geq 5$  plays the role of agent 2, 3 or 4, then no improvement can be applied as i ranks e at the top. Finally, neutrality can be recovered as above.

# **B** Proof of Theorem 2

We begin by introducing some additional notation. Given  $R_i \in \mathcal{R}^i$ , let  $top(R_i) \in O$  denote the top-ranked object in O according to  $R_i$ , i.e.,  $top(R_i)R_ix$  for all  $x \in O$ . For a subset  $I' \subseteq N$ , let  $top(R_{I'}) = \bigcup_{i \in I'} \{top(R_i)\}$  and denote the set of objects top-ranked by some  $i \in N$  by  $top(R) = \bigcup_{i \in N} \{top(R_i)\}$ . Conversely, let  $\overline{top}(R) = O \setminus top(R)$  denote the set of objects which are not top-ranked by any  $i \in N$ .

If all agents rank a different object at the top, i.e., if top(R) = O, Pareto efficiency requires that each agent receives their top-ranked object. Our first lemma concerns an implication of efficiency when preferences are at least partially in conflict, i.e., if  $top(R) \neq O$  – top-ranked objects will not be assigned to agents who rank them at the bottom, i.e., below non-top ranked objects. **Lemma 2** Consider any ex-post efficient mechanism g and any preference profile  $R \in \mathbb{R}^N$ such that  $top(R) \neq O$ . Then for all  $i \in N$  and all  $y \in top(R)$  such that  $xP_iy$  for all  $x \in \overline{top}(R)$ , we have  $g_{iy}(R) = 0$ . Moreover, for any mechanism f such that  $f \triangleright^{sd} g$ , we also have  $f_{iy}(R) = 0$ .

**Proof.** Towards a contradiction, assume there exists an  $i \in N$  and  $y \in top(R) \subsetneq O$ , such that  $g_{iy}(R) > 0$  while for all  $x \in \overline{top}(R)$  we have  $xP_iy$ . Since g is ex-post efficient, there exists  $\mu \in \mathcal{PO}(R)$  such that  $\mu_i = y$ .

Since  $top(R_i)P_ixP_i\mu_i$  for all  $x \in \overline{top}(R)$ , we have  $top(R_i) \neq \mu_i$  – and since  $\mu_i \in top(R)$ , there must be another agent,  $j \in N \setminus \{i\}$ , for whom  $top(R_j) = \mu_i$ .

But then  $top(R_j) \neq \mu_j$ . By efficiency,  $\mu_j \in top(R)$  – as otherwise  $\mu_j P_i \mu_i$  and  $\mu_i = top(R_j)P_j\mu_j$ , creating a possible trading cycle where all included agents become strictly better off. Thus, there must be another agent,  $k \in N \setminus \{i, j\}$ , for whom  $top(R_k) = \mu_j$ .

But then  $top(R_k) \neq \mu_k$ . By efficiency,  $\mu_k \in top(R)$  – as otherwise  $\mu_k P_i \mu_i$ ,  $\mu_i P_j \mu_j$ , and  $\mu_j P_k \mu_k$ , creating a possible trading cycle where all included agents become strictly better off. Thus there must be another agent,  $l \in N \setminus \{i, j, k\}$  for whom  $top(R_l) = \mu_j$ .

Continue in this way. Since N is finite, we eventually arrive at a contradiction once we have exhausted N. This establishes the first part of the lemma: no agent receives objects ranked below their least preferred object in  $\overline{top}(R)$  under g. A fortiori, the same needs to hold for any mechanism f which stochastically dominates g.

Suppose now that mechanism g satisfies ex-post efficiency, bounded invariance and strategyproofness. Towards a contradiction, assume there exists a bounded invariant and strategyproof mechanism f such that  $f \triangleright^{sd} g$ . In particular, this implies that there is a non-empty set of preference profiles where f and g are not welfare-equivalent. Let  $\mathcal{R}_0^{\neq}$  denote this set, i.e.,

$$\mathcal{R}_0^{\neq} = \{ R \in \mathcal{R}^N : f_i(R) \neq g_i(R) \text{ for some } i \in N \}.$$

To prove Theorem 2 by contradiction, we will show that  $\mathcal{R}_0^{\neq} = \emptyset$ . For this, we will consider an arbitrary sequence of objects  $z_1, z_2, \ldots, z_n \in O$  along with a decreasing sequence of subsets of preference profiles

$$\mathcal{R}_0^{
eq} \supseteq \mathcal{R}_1^{
eq} \supseteq \mathcal{R}_2^{
eq} ... \supseteq \mathcal{R}_n^{
eq}$$

where (i) for each k = 1, ..., n,  $\mathcal{R}_{k-1}^{\neq} \neq \emptyset$  implies  $\mathcal{R}_{k}^{\neq} \neq \emptyset$  while (ii)  $f_{z_{l}}(R) = g_{z_{l}}(R)$  for all  $l \leq k$  and all  $R \in \mathcal{R}_{k}^{\neq}$ . This way, (i) implies  $\mathcal{R}_{n}^{\neq} \neq \emptyset$  (given  $\mathcal{R}_{0}^{\neq} \neq \emptyset$ ), while (ii) implies that for  $R \in \mathcal{R}_{n}^{\neq}$  all objects  $z_{1}, ..., z_{n}$  have to be assigned with the same assignment probabilities under f and g – i.e.,  $\mathcal{R}_{n}^{\neq} = \emptyset$ .

Intuitively, each  $\mathcal{R}_k^{\neq}$  is the set of preference profiles where  $z_k$  is ranked as low as possible by all agents (relative to objects in  $\overline{top}(R)$ ), subject to the constraint that  $\mathcal{R}_k^{\neq} \subseteq \mathcal{R}_{k-1}^{\neq} \subseteq$  $\dots \subseteq \mathcal{R}_0^{\neq}$ , i.e., subject to preserving a difference between f and g and subject to ranking the preceding objects  $z_{k-1}, z_{k-2}..., z_1$  as low as possible. Moreover, note that for all preference profiles in  $\mathcal{R}_0^{\neq}$ , and hence also for all profiles in  $\mathcal{R}_k^{\neq}$ , we have  $top(R) \neq O$  as otherwise ex-post efficiency requires that all agents receive their top ranked object with probability one so that f(R) = g(R).

To make this precise and define the sets  $\mathcal{R}_k^{\neq}$  formally, let  $\mathbb{N} = \{0, 1, \ldots\}$  denote the set of natural numbers including zero. Let  $\mathbb{N}_{\geq}^{|N|}$  denote the set of all vectors  $v \in \mathbb{N}^{|N|}$  such that  $v_1 \geq v_2 \geq \cdots \geq v_{|N|}$ , i.e., the coordinates of v are arranged in non-increasing order. Let  $\preceq$  denote the lexicographical ordering on  $\mathbb{N}_{\geq}^{|N|}$ : for all  $v, w \in \mathbb{N}_{\geq}^{|N|}$ ,  $v \preceq w$  means either v = w or there is  $1 \leq t \leq n$ , such that  $v_t < w_t$  and  $v_i = w_i$  for every i < t and . We write  $v \prec w$  if  $v \preceq w$  and  $w \neq v$ .

Furthermore, for any  $z \in O$  and  $R_i$ , let

$$L(z, R_i) = \{ y \in O : zP_i y \}$$

denote the strict lower contour set of z at  $R_i$ . Note that this set excludes z.

Next, let  $O = \{z_1, \ldots, z_n\}$ , define  $O_{\geq t} = \{z_t, \ldots, z_n\}$ , for any  $1 \leq t < n$ , and for any  $i \in N$  let

$$\rho_i(z_t, R) = |L(z_t, R_i) \cap O_{\ge t} \cap \overline{top}(R)|$$

be the rank that  $z_t$  occupies in agents' preferences, where the rank of  $z_t$  is the number of non-top-ranked objects below  $z_t$ , ignoring objects  $z_l$  with l < t. Further, let  $\theta(z_t, R) \in \mathbb{N}^n_{\geq}$ be the vector of ranks, ordered in non-increasing fashion, i.e.,  $\theta_i(z_t, R) = \rho_{\tau(i)}(z_t, R)$  for an appropriate permutation  $\tau : N \to N$ . For any  $t \geq 1$  define

$$\mathcal{R}_t^{\neq} = \{ R \in \mathcal{R}_{t-1}^{\neq} : \text{there exists no } \bar{R} \in \mathcal{R}_{t-1}^{\neq} \text{ such that } \theta(z_t, \bar{R}) \prec \theta(z_t, R) \},\$$

where  $\theta(z_t, \bar{R})$  and  $\theta(z_t, R)$  are ordered by lexicographic minimization. Hence,  $\mathcal{R}_t^{\neq}$  contains all profiles where  $z_t$  is ranked as low as possible, provided that (i) f and g still differ in the assignment probability shares of *some* object, and that (ii) all objects  $z_l \in O$ , l < t, are likewise ranked as low as possible (with rank-minimization of  $z_m$  taking precedence over the rank-minimization of  $z_{m'}$  for any m < m' < t).

We first show in two lemmas that for any preference profile in  $\mathcal{R}_1^{\neq}$ , the assignment probabilities of  $z_1$  coincide under f and g. We then proceed by induction to show that the same holds for any  $\mathcal{R}_k^{\neq}$ , and objects  $z_l, l \leq k$ . This implies that  $\mathcal{R}_n^{\neq} = \emptyset$  and thus establishes the desired contradiction.

**Lemma 3** Consider  $z_1 \in O$  and  $R \in \mathcal{R}_1^{\neq}$ , and partition N as follows:  $N = I_1 \cup I_2$  with  $I_1 = \{i \in N : L(z_1, R_i) \cap \overline{top}(R) = \emptyset\}$  and  $I_2 = N \setminus I_1$  (i.e.,  $I_1$  consists of those agents for which  $z_1$  is ranked least relative to  $\overline{top}(R)$  while agents in  $I_2$  rank some object from  $\overline{top}(R)$  below  $z_1$ ). If there is some  $j \in N$  such that  $f_{jz_1}(R) > g_{jz_1}(R)$ , then  $j \in I_2$  and for all  $i \in I_2 \setminus \{j\}$  we have  $(L(z_1, R_i) \cap \overline{top}(R)) \supseteq (L(z_1, R_j) \cap \overline{top}(R)) \neq \emptyset$  (i.e., i's lower contour set of  $z_1$  at  $R_i$  contains all objects in  $\overline{top}(R)$  which are contained in j's lower contour set).

**Proof.** First, given that  $f \triangleright^{sd} g$ ,  $f_{jz_1}(R) > g_{jz_1}(R)$  implies that there is some object x ranked below  $z_1$  by j, for which  $f_{jx}(R) < g_{jx}(R)$ . If  $x \in \overline{top}(R)$ , then we have  $L(z_1, R_j) \cap \overline{top}(R) \neq \emptyset$ ; and if  $x \in top(R)$ , then Lemma 2 implies  $L(z_1, R_j) \cap \overline{top}(R) \subseteq L(x, R_j) \cap \overline{top}(R) \neq \emptyset$ . Hence, in either case we have  $j \in I_2$ . Now take any  $a \in L(z_1, R_j) \cap \overline{top}(R)$  and move it up to just below  $z_1$ , arriving at R'. Note that top(R) = top(R') so that  $\theta(z_1, R) = \theta(z_1, R')$  and hence  $R' \in \mathcal{R}_1^{\neq}$ . By strategy-proofness we have  $f_{jz_1}(R') > g_{jz_1}(R')$  and since  $R' \in \mathcal{R}_1^{\neq}$  we have  $f_{ja}(R') < g_{ja}(R')$  – otherwise, we could swap a and  $z_1$ , arriving at R'' where  $f(R'') \neq g(R'')$ , yet  $top(R'') \setminus \{z_1\} \supseteq top(R') \setminus \{z_1\}$  so that  $z_1$  is ranked lower relative to non-top-ranked objects in R'' than in R', contradicting  $R' \in \mathcal{R}_1^{\neq}$ . Any agent  $i \neq j$  who does not rank  $z_1$  least relative to  $\overline{top}(R') = \overline{top}(R)$ , i.e., for whom  $L(z_1, R_j) \cap \overline{top}(R') \neq \emptyset$ , must also rank a below  $z_1$  in  $R_i = R'_i$ : otherwise, they could move  $z_1$  to the bottom of their preferences in R' – call the new profile R'''. By BI, we still have  $f_{ja}(R''') < g_{ja}(R''')$ . Again, this would contradict  $R' \in \mathcal{R}_1^{\neq}$ , i.e., that  $z_1$  is ranked as low as possible in R'.

Since  $a \in L(z_1, R_j) \cap \overline{top}(R)$  was chosen arbitrarily, this proves the Lemma 3.

**Lemma 4** Let  $R \in \mathcal{R}_1^{\neq}$ . Then  $f_{iz_1}(R) = g_{iz_1}(R)$  for all  $i \in N$ .

**Proof.** Towards a contradiction, assume there exists  $j \in N$  with  $f_{jz_1}(R) > g_{jz_1}(R)$  and consider the partition  $\{I_1, I_2\}$  as in Lemma 3. By Lemma 3 we know that  $j \in I_2$ , i.e.,  $L(z_1, R_j) \cap \overline{top}(R) \neq \emptyset$ , and  $0 = \rho_h(z_1, R) < \rho_j(z_1, R) \leq \rho_l(z_1, R)$  for all  $h \in I_1$  and  $l \in I_2$ . We will construct a new profile  $\tilde{R}^*$  such that  $f(\tilde{R}^*) \neq g(\tilde{R}^*)$  but where the number of nontop ranked objects below  $z_1$  for agents in  $I_2$  is lower than in R – strictly so for  $j \in I_2$  – and where for all agents in  $i \in I_1$  we have  $\rho_i(z_1, \tilde{R}^*) < \rho_j(z_1, R)$ . Thereby we will find that  $\theta(z_1, \tilde{R}^*) \prec \theta(z_1, R)$ , contradicting  $R \in \mathcal{R}_1^{\neq}$ .

First we will rule out  $z_1 \in top(R)$ . For that, note that since  $R \in \mathcal{R}_1^{\neq}$ , we need to have  $f_{iz_1}(R) \geq g_{iz_1}(R)$  for all  $i \in I_2$  – otherwise, for  $i \in I_2$  such that  $f_{iz_1}(R) < g_{iz_1}(R)$ ,  $f \triangleright^{sd} g$  would imply there to be a higher-ranked object,  $xP_iz_1$ , such that  $f_{ix}(R) > g_{ix}(R)$  and we could move  $z_1$  to the bottom of i preference order. For the new profile, denoted  $\hat{R}$ , strategy-proofness would imply  $f_{ix}(\hat{R}) > g_{ix}(\hat{R})$ , i.e.,  $f(\hat{R}) \neq g(\hat{R})$ . Since  $top(R) = top(\hat{R})$  and  $z_1$  is now ranked

lower for *i* but unchanged for all  $k \neq i$ , i.e.,  $0 = |L(z_t, \hat{R}_i) \cap \overline{top}(\hat{R})| < |L(z_t, R_i) \cap \overline{top}(R)|$ and  $|L(z_t, \hat{R}_k) \cap \overline{top}(\hat{R})| = |L(z_t, R_k) \cap \overline{top}(R)|$ , this contradicts  $R \in \mathcal{R}_1^{\neq}$ . We conclude that  $f_{iz_1}(R) \ge g_{iz_1}(R)$ , for all  $i \in I_2$ .

But then it cannot be the case that  $z_1 \in top(R)$ , since Lemma 2 would imply  $f_{iz_1}(R) = 0 = g_{iz_1}(R)$ , for all  $i \in I_1$ , which, together with  $f_{iz_1}(R) \ge g_{iz_1}(R)$ , for all  $i \in I_2$ , as well as  $f_{jz_1}(R) > g_{jz_1}(R)$ , would contradict the fact that  $z_1$  is assigned with probability one in both f and g.

Now, if  $top(R_{I_1}) \cap L(z_1, R_j) \neq \emptyset$ , take any  $x \in top(R_{I_1}) \cap L(z_1, R_j)$  and move up xin  $R_j$  just below  $z_1$  to arrive at  $R_j^x$ . Note that  $top(R_j^x, R_{-j}) = top(R)$  and  $L(z_1, R_j^x) = L(z_1, R_j)$ . By strategy-proofness, we still have  $f_{jz_1}(R_j^x, R_{-j}) > g_{jz_1}(R_j^x, R_{-j})$ . We have either  $f_{jx}(R_j^x, R_{-j}) < g_{jx}(R_j^x, R_{-j})$  or  $f_{jx}(R_j^x, R_{-j}) \ge g_{jx}(R_j^x, R_{-j})$ . We show that for both cases we obtain a new profile R' where  $f_{jz_1}(R') > g_{jz_1}(R')$ , where  $\rho_i(z_1, R') = \rho_i(z_1, R)$  for all  $i \in I_2$  and where  $\rho_i(z_1, R') \le \rho_j(z_1, R)$  for all  $i \in I_1$ . Let  $I_1^x$  denote the set of agents in  $I_1$ who rank x at the top.

Case (1.x): if  $f_{jx}(R_j^x, R_{-j}) < g_{jx}(R_j^x, R_{-j})$ , let all  $i \in I_1^x$  push  $\{z_1\} \cup (L(z_1, R_j) \cap \overline{top}(R))$  to the bottom of their preference order, in the same order as they are ranked in  $R_j$ , to arrive at  $R'_i$ . For j, relabel  $R'_j = R^x_j$  and for all other  $i \in N \setminus (I_1^x \cup \{j\})$ , relabel  $R'_i = R_i$  to arrive at  $R' = (R'_k)_{k \in N}$ . By BI, we still have  $f_{jx}(R') < g_{jx}(R')$ . Towards a contradiction, assume  $f_{jz_1}(R') \leq g_{jz_1}(R')$ . Then there would be some object  $yP'_jz_1$  such that  $f_{jy}(R') > g_{jy}(R')$ . Moreover,  $yP'_iz_1$  for all  $i \in I_1^x$ . Hence we could push  $z_1$  to the bottom of the preference order for all agents  $i \in I_1^x$  as well as for j and, by BI, arrive at a profile  $\hat{R}$  where f and gdiffer in the assignment probabilities of y. Since in  $\hat{R}$ ,  $z_1$  is ranked lower relative to objects  $\overline{top}(\hat{R}) = \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_1^{\neq}$  – and we conclude that  $f_{jz_1}(R') > g_{jz_1}(R')$ .

Case (2.x): if instead we have  $f_{jx}(R_j^x, R_{-j}) \ge g_{jx}(R_j^x, R_{-j})$ , swap x and  $z_1$  in the ranking of j – let us denote this new preference order as  $R'_j$  and the new preference profile  $(R'_j, R_{-j})$  simply as R'. Note that since  $z_1 \ne top(R_j^x) = top(R_j)$  we have top(R') = top(R). Towards a contradiction, assume  $f_{jx}(R') > g_{jx}(R')$ . Then we could push down  $z_1$  to the bottom of j's preference order, below all other  $\overline{top}(R')$ , and do the same for all  $i \in I_1^x$  – call the new preference profile  $\hat{R}$ . By BI this preserves  $f_{jx}(\hat{R}) > g_{jx}(\hat{R})$ . Since in  $\hat{R}$  object  $z_1$  is ranked lower relative to objects  $\overline{top}(\hat{R}) = \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_1^{\neq}$ . Therefore, after having swapped x and  $z_1$ , we must have  $f_{jx}(R') \le g_{jx}(R')$  and thus  $f_{jz_1}(R') > g_{jz_1}(R')$ .

Thus, independently of whether Case (1.x) or Case (2.x) applies, we arrive at a new profile R' where  $f_{jz_1}(R') > g_{jz_1}(R')$  and where  $\rho_i(z_1, R') = \rho_i(z_1, R)$  for all  $i \in I_2$ , i.e.,  $z_1$  is ranked as low as before for all agents in  $I_2$ . While  $z_1$  might be ranked higher than in R for agents in  $I_1^x$ , we still have  $\rho_k(z_1, R') \leq \rho_j(z_1, R') \leq \rho_l(z_1, R')$  for all  $k \in I_1$  and  $l \in I_2$ .

Next, if there is any other  $x' \in (top(R_{I_1}) \cap L(z_1, R_j)) \setminus \{x\} \subseteq top(R'_{I_1}) \cap L(z_1, R'_j)$ , we proceed as before and move up x' in  $R'_j$  just below  $z_1$ . Refer to this preference order as  $R''_j$ . By strategy-proofness, we still have  $f_{jz_1}(R''_j, R'_{-j}) > g_{jz_1}(R''_j, R'_{-j})$ . We proceed as above and obtain profile R'' where  $f_{jz_1}(R'') > g_{jz_1}(R'')$  and the rank of  $z_1$  relative to non-top-ranked objects remains unchanged for agents in  $I_2$ .

Case (1.x'): if  $f_{jx'}(R_j^{x'}, R_{-j}') < g_{jx'}(R_j^{x'}, R_{-j}')$  we proceed as in Case (1.x) – the only difference is that we now need to take into account the possible changes made to preferences of agents in  $I_1^x$  in Case (1.x). Let all  $i \in I_1^{x'}$  push  $\{z_1\} \cup (L(z_1, R_j) \cap \overline{top}(R))$  to the bottom of their preference order, in the same order as they are ranked in  $R_j$ , to arrive at  $R_i''$ . For j, relabel  $R_j'' = R_j^{x'}$  and for all other  $i \in N \setminus (I_1^{x'} \cup \{j\})$ , relabel  $R_i'' = R_i'$  to arrive at  $R'' = (R_k'')_{k \in N}$ . By BI, we still have  $f_{jx'}(R'') < g_{jx'}(R'')$ . Towards a contradiction, assume  $f_{jz_1}(R'') \leq g_{jz_1}(R'')$ . Then there would be some object  $yP_j''z_1$  such that  $f_{jy}(R'') > g_{jy}(R'')$ . Moreover,  $yP_i''z_1$ for all  $i \in I_1^{x'}$  as well as for all  $i \in I_1^x$  if we arrived at R' via Case (1.x). Hence we could push  $z_1$  to the bottom of the preference order for all agents in  $I_1$  for whom we have so far constructed new preferences<sup>33</sup> as well as for j and, by BI, arrive at a profile  $\hat{R}$  where f and gdiffer in the assignment probabilities of y. Since in  $\hat{R}$ ,  $z_1$  is ranked lower relative to objects  $\overline{top}(\hat{R}) = \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_1^{\neq}$  – and we conclude that  $f_{jz_1}(R'') > g_{jz_1}(R'')$ .

Case (2.x'): if instead we have  $f_{jx'}(R_j^{x'}, R_{-j}') \ge g_{jx'}(R_j^{x'}, R_{-j}')$ , swap x' and  $z_1$  in the ranking of j – let us denote this new preference order as  $R_j''$  and the new preference profile  $(R_j'', R_{-j}')$ simply as R''. Note that since  $z_1 \ne top(R_j^x) = top(R_j)$  we have top(R'') = top(R). Towards a contradiction, assume  $f_{jx}(R'') > g_{jx}(R'')$ . Then we could push down  $z_1$  to the bottom of j's preference order, below all other  $\overline{top}(R'')$ , and do the same for all  $i \in I_1^{x'}$ , as well as for all other  $i \in I_1$  for whom we may have so far constructed new preferences – call the new preference profile  $\hat{R}$ . By BI this preserves  $f_{jx}(\hat{R}) > g_{jx}(\hat{R})$ . Since in  $\hat{R}$  object  $z_1$  is ranked lower relative to objects  $\overline{top}(\hat{R}) = \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_1^{\neq}$ . Therefore, after having swapped x and z, we must have  $f_{jx}(R'') \le g_{jx}(R'')$  and thus  $f_{jz_1}(R'') > g_{jz_1}(R'')$ .

Repeat these steps for all  $x^* \in top(R_{I_1}) \cap L(z_1, R_j)$ , i.e., move up  $x^*$  in the preference order of j to just below  $z_1$  and then proceed as in Case (1.x') or (2.x'). This way, we arrive at a profile, refer to it as  $R^{\dagger}$ , where  $top(R_i^{\dagger}) = top(R_i)$  for all  $i \in N$ ,  $f_{jz_1}(R^{\dagger}) > g_{jz_1}(R^{\dagger})$ , and  $I_1$  has been partitioned into two subsets:  $I'_1$  includes all agents  $i \in I_1$  for whom  $R_i^{\dagger} = R_i$ and hence  $L(z_1, R_i^{\dagger}) \cap \overline{top}(R) = \emptyset$ , and whose top ranked objects are ranked above  $z_1$  by  $j = I_1$ .

<sup>&</sup>lt;sup>33</sup>I.e., for  $i \in I_1^{x'} \cup I_1^x$  if we arrived at R' via Case (1.x), and for  $i \in I_1^{x'}$  if we arrived at R via Case (2.x).

in R but also in  $R^{\dagger}$  since j's lower contour set has only gotten weakly smaller as we moved away from R to  $R^{\dagger}$  (strictly smaller whenever Case 2 applied). Second,  $I''_1$  includes all agents  $i \in I_1$  whose lower contour set  $L(z_1, R_i^{\dagger})$  consists of all objects  $L(z_1, R_j) \cap \overline{top}(R)$ . Third, compared to R, j's lower contour set at  $z_1$  has gotten weakly smaller in that some objects from  $top(R_{I_1})$  may now be ranked above  $z_1$  – however, no object in  $\overline{top}(R)$  has been raised above  $z_1$  as we moved to  $R_j^{\dagger}$ , i.e.,  $L(z_1, R_j) \cap \overline{top}(R) = L(z_1, R_j^{\dagger}) \cap \overline{top}(R^{\dagger})$ . Last the ranking of other agents  $i \in I_2 \setminus \{j\}$  is unchanged, i.e.,  $R_i^{\dagger} = R_i$ .

By Lemma 3 as well as the preceding construction, we have for all  $h \in I_1''$  and all  $l \in I_2$ ,

$$L(z_1, R_h^{\dagger}) \cap \overline{top}(R^{\dagger}) \subseteq L(z_{k+1}, R_j^{\dagger}) \cap \overline{top}(R^{\dagger}) \subseteq L(z_1, R_l^{\dagger}) \cap \overline{top}(R^{\dagger}),$$
  

$$\rho_h(z_1, R^{\dagger}) \leq \rho_j(z_1, R^{\dagger}) \leq \rho_l(z_1, R^{\dagger}) \text{ and } \rho_l(z_1, R^{\dagger}) = \rho_l(z_1, R).$$

Now, for all  $i \in I_1'' \cup I_2$  (including j) change the order of objects in the lower contour set  $L(z_1, R_i^{\dagger})$  as follows: (i) objects that are in  $L(z_1, R_i^{\dagger}) \setminus L(z_1, R_j^{\dagger})$  are ranked immediately below  $z_1$  (beyond that, their order does not matter), (ii) objects that are also in  $L(z_1, R_j^{\dagger}) \cap \overline{top}(R^{\dagger})$  are ranked next, in the same order as by  $R_j^{\dagger}$ , (iii) last, all objects in  $L(z_1, R_i^{\dagger}) \cap L(z_1, R_j^{\dagger}) \cap top(R^{\dagger})$  or  $top(R^{\dagger})$  are ranked below (beyond that, their order does not matter). Call this new (and penultimate) profile  $\tilde{R}$ . By BI, we still have  $f_{jz_1}(\tilde{R}) > g_{jz_1}(\tilde{R})$ . By Lemma 2 and  $f \triangleright^{sd} g$ , we have  $f_{ix}(\tilde{R}) = 0 = g_{ix}(R)$  for all  $i \in I_2$  and all  $x \in L(z_1, \tilde{R}_i) \cap L(z_1, \tilde{R}_j) \cap top(\tilde{R})$ .

Hence, we now have all agents in  $I''_1 \cup I_2$  ranking objects  $L(z_1, \tilde{R}_j) \cap \overline{top}(\tilde{R})$  adjacent and in the same order as  $\tilde{R}_j$ , and below that only objects in  $top(\tilde{R}) = top(R)$  for which the assignment probabilities are equal to zero under f and g by Lemma 2. Since  $f_{jz_1}(\tilde{R}) > g_{jz_1}(\tilde{R})$ , there is some y, ranked below  $z_1$  by  $\tilde{R}_j$ , such that  $f_{jy}(\tilde{R}) < g_{jy}(\tilde{R})$  – and thus some  $i \in N$  with  $f_{iy}(\tilde{R}) > g_{iy}(\tilde{R})$ . Moreover, by Lemma 2, we have  $y \in L(z_1, \tilde{R}_j) \cap \overline{top}(\tilde{R})$ .

If  $i \in I''_1 \cup I_2$ , then there is y' with  $y\tilde{R}_iy'$ , such that  $f_{iy'}(\tilde{R}) < g_{iy'}(\tilde{R})$  – and thus some  $i' \in N$  with  $f_{i'y'}(\tilde{R}) > g_{i'y'}(\tilde{R})$ . Hence, by Lemma 2, it must be that  $y \in \overline{top}(\tilde{R})$ , so that  $y' \in L(y, \tilde{R}_j) \cap \overline{top}(\tilde{R})$ . Thus, y' is ranked lower than y according to  $\tilde{R}_j$ .

If  $i' \in I_1'' \cup I_2$ , then there is y'' with  $y'\tilde{R}_{i'}y''$ , such that  $f_{i'y''}(\tilde{R}) < g_{i'y''}(\tilde{R})$  – and thus some  $i'' \in N$  with  $f_{i''y''}(\tilde{R}) > g_{i''y''}(\tilde{R})$ , and so on.

Since  $L(z, \tilde{R}_j) \cap \overline{top}(\tilde{R})$  is finite and we move down (according to  $\tilde{R}_j$ ) in each iteration, eventually there is some  $y^* \in L(z, \tilde{R}_j) \cap \overline{top}(\tilde{R})$  and  $i^* \in I'_1 = N \setminus (I''_1 \cup I_2)$  such that  $f_{i^*y^*}(\tilde{R}) > g_{i^*y^*}(\tilde{R}).$ 

Note that  $\tilde{R}_i = R_i$ , and thus,  $y^* \tilde{P}_i z_1$  for any  $i \in I'_1$ . If  $y^* P_i top(\tilde{R}_{i^*})$ , then change  $\tilde{R}_i$  to  $\tilde{R}'_i$  as follows: (i) objects in  $B(y^*, R_i)$  are ranked first according to  $R_i$ , (ii) then  $top(\tilde{R}_{i^*})$  and (iii) then objects in  $L(y^*, R_i) \setminus \{top(\tilde{R}_{i^*})\}$  according to  $R_i$ . After having done this for all such  $i \in I'_1$  and denoting the obtained profile by  $\tilde{R}'$ , by BI we continue to have  $f_{i^*y^*}(\tilde{R}') > g_{i^*y^*}(\tilde{R}')$ .

But then let  $i^*$  exchange the positions of  $y^*$  and  $top(\tilde{R}_{i^*})$  in  $\tilde{R}'_{i^*}$  and call this final profile  $\tilde{R}^*$ . This strictly decreases the number of non-top objects ranked below  $z_1$  for j, as well as all  $i \in I''_1$ , and weakly decreases it for all  $i \in I'_1$  (as either  $top(\tilde{R}_{i^*})\tilde{P}'_iy^*\tilde{P}'_iz_1$  or  $top(\tilde{R}_{i^*})$  is ranked immediately below  $y^*$  in  $\tilde{R}'_i$ ) and for all  $i \in I_2 \setminus \{j\}$  (only weakly if  $i \in I_2 \setminus \{j\}$  ranked both  $top(\tilde{R}_{i^*})$  and  $top(\tilde{R}^*_{i^*})$  below  $z_1$ ). Hence,  $\rho_i(z_1, \tilde{R}^*) \leq \rho_i(z_1, R)$  for  $i \in I_2 \setminus \{j\}$ ,  $\rho_j(z_1, \tilde{R}^*) < \rho_j(z_1, \tilde{R})$ , and  $\rho_i(z_1, \tilde{R}^*) \leq \rho_j(z_1, \tilde{R}^*)$  for  $i \in I_1$  contradicting  $R \in \mathcal{R}_1^{\neq}$ .

The following two lemmas extend Lemma 3 and Lemma 4 to  $\mathcal{R}_t^{\neq}$ , t = 1, ..., n, thereby completing the proof. Recall that  $O_{\geq t} = \{z_t, ..., z_n\}$ , for any  $1 \leq t < n$ . Let  $Z_t = \{z_1, ..., z_t\}$  for any  $1 \leq t < n$ .

**Lemma 5** Consider  $1 \le t < n$ ,  $z_t \in O$  and  $R \in \mathcal{R}_t^{\neq}$ , and partition N as follows:  $N = I_1 \cup I_2$ with  $I_1 = \{i \in N : L(z_t, R_i) \cap O_{\ge t} \cap \overline{top}(R) = \emptyset\}$  and  $I_2 = N \setminus I_1$  (i.e.,  $I_1$  consists of those agents for which  $z_t$  is ranked least relative to  $O_{\ge t} \cap \overline{top}(R)$  while agents in  $I_2$  rank some object from  $O_{\ge t} \cap \overline{top}(R)$  below  $z_t$ ). If there is some  $j \in N$  such that  $f_{jz_t}(R) > g_{jz_t}(R)$  then  $j \in I_2$ and for all  $i \in I_2 \setminus \{j\}$  we have  $L(z_t, R_i) \cap O_{\ge t} \cap \overline{top}(R) \supseteq L(z_t, R_j) \cap O_{\ge t} \cap \overline{top}(R) \neq \emptyset$  (i.e., i's lower contour set of  $z_t$  at  $R_i$  contains all objects in  $O_{\ge t} \cap \overline{top}(R)$  which are contained in j's lower contour set).

**Lemma 6** Let  $R \in \mathcal{R}_t^{\neq}$ . Then  $f_{iz_t}(R) = g_{iz_t}(R)$  for all  $i \in N$ .

**Proof of Lemma 5 and 6.** For t = 1, this is established by Lemma 3 and 4, which serve as the basis for the following induction. For the induction step, assume we have established both statements for all  $1 \le t \le k < n$  (the induction hypothesis). It remains to show that both hold for t = k + 1. For this, the following observation will be useful.

Claim 6 If there is some  $j \in I$  such that  $f_{jz_{k+1}}(R) > g_{jz_{k+1}}(R)$  for  $R \in \mathcal{R}_{k+1}^{\neq}$ , then  $z_{k+1} \neq top(R_j)$ .

Proof of Claim 6: Suppose  $top(R_j) = z_{k+1}$ . Since  $f_{jz_{k+1}}(R) > g_{jz_{k+1}}(R)$ , there exists i with  $f_{iz_{k+1}}(R) < g_{iz_{k+1}}(R)$ . We have either  $L(z_{k+1}, R_i) \cap O_{\geq k+1} \cap \overline{top}(R) \neq \emptyset$  or  $L(z_{k+1}, R_i) \cap O_{\geq k+1} \cap \overline{top}(R) = \emptyset$ .

Suppose  $L(z_{k+1}, R_i) \cap O_{\geq k+1} \cap \overline{top}(R) \neq \emptyset$ . By  $f \rhd^{sd} g$  there exists x with  $xP_i z_{k+1}$ and  $f_{ix}(R) > g_{ix}(R)$ . But then we could move  $Z_{k+1} \cap L(z_{k+1}, R_i)$  to the bottom of  $R_i$  in unchanged order, arriving at a contradiction to  $R \in \mathcal{R}_{k+1}^{\neq}$ .

Suppose instead  $L(z_{k+1}, R_i) \cap O_{\geq k+1} \cap \overline{top}(R) = \emptyset$ . By Lemma 2 and  $z_{k+1} \in top(R)$ , there exists  $y \in \overline{top}(R) \cap L(z_{k+1}, R_i)$ . Hence we must have  $y \in Z_k$ . Furthermore, by  $f \triangleright^{sd} g$  and  $f_{iz_{k+1}}(R) < g_{iz_{k+1}}(R)$ , we have  $\sum_{o \in O:oP_i z_{k+1}} f_{io}(R) > \sum_{o \in O:oP_i z_{k+1}} g_{io}(R)$ . Now reorder the

objects in  $R_i$  as follows: first rank  $O \setminus (L(z_{k+1}, R_i) \cup \{z_{k+1}\})$ , then  $\overline{top}(R) \cap L(z_{k+1}, R_i)$ , and then, at the bottom  $top(R) \cap L(z_{k+1}, R_i) \cup \{z_{k+1}\}$ . Call the new preference profile R'. Since objects  $O \setminus (L(z_{k+1}, R_i) \cup \{z_{k+1}\})$  are still ranked above objects in  $(L(z_{k+1}, R_i) \cup \{z_{k+1}\})$ , strategy-proofness implies  $\sum_{o \in O \setminus (L(z_{k+1}, R_i) \cup \{z_{k+1}\})} f_{io}(R') > \sum_{o \in O \setminus (L(z_{k+1}, R_i) \cup \{z_{k+1}\})} g_{io}(R')$ . Yet by the induction hypothesis and Lemma 2, we have  $f_{iy}(R') = g_{iy}(R')$  for all  $y \in$  $(L(z_{k+1}, R_i) \cup \{z_{k+1}\})$ . Hence  $\sum_{o \in O} f_{io}(R') > \sum_{o \in O} g_{io}(R')$  – contradicting feasibility and thus establishing Claim 6.

Induction step for Lemma 5. First, given that  $f \triangleright^{sd} g$ ,  $f_{jz_{k+1}}(R) > g_{jz_{k+1}}(R)$  implies that there is some object x ranked below  $z_{k+1}$  for which  $f_{jx}(R) < g_{jx}(R)$ . Thus, by Lemma 2 and the induction hypothesis, we have  $L(z_{k+1}, R_j) \cap \overline{top}(R) \cap O_{\geq k+1} \neq \emptyset$ , i.e.,  $j \in I_2$ : otherwise all objects in  $L(z_{k+1}, R_j)$  would either be top-ranked objects or from  $Z_k$ . But then we could move to  $R'_j$  by reordering j's lower contour set, pushing all objects in top(R) to the bottom. Since this leaves the rank of objects  $z \in Z_{k+1}$  unaffected, we would still have  $(R'_j, R_{-j}) \in \mathcal{R}_{k+1}^{\neq}$ . But now Lemma 2 implies that  $f_{jx}(R'_j, R_{-j}) = g_{jx}(R'_j, R_{-j})$  for all  $x \in top(R) = top(R'_j, R_{-j})$ while the induction hypothesis implies that  $f_{jx}(R'_j, R_{-j}) = g_{jx}(R'_j, R_{-j})$  for all  $x \in Z_k$ . Since strategy-proofness ensures  $f_{jz_{k+1}}(R'_j, R_{-j}) > g_{jz_{k+1}}(R'_j, R_{-j})$ , this would contradict  $f \triangleright^{sd} g$ .

Now take any  $a \in L(z_{k+1}, R_j) \cap O_{\geq k+1} \cap \overline{top}(R)$  and move it up to just below  $z_{k+1}$ , arriving at R'. Note that top(R) = top(R') so that  $\rho(z_{k+1}, R) = \rho(z_{k+1}, R')$  and hence  $R' \in \mathcal{R}_{k+1}^{\neq}$ . By strategy-proofness we have  $f_{jz_{k+1}}(R') > g_{jz_{k+1}}(R')$  and since  $R' \in \mathcal{R}_{k+1}^{\neq}$ we have  $f_{ja}(R') < g_{ja}(R')$  – otherwise, we could swap a and  $z_{k+1}$  in  $R'_j$ , arriving at R''where  $f_l(R'') \neq g_l(R'')$  for some  $l \in N$ , yet, by Claim 6, top(R'') = top(R'), so that  $z_{k+1}$  is ranked lower relative to non-top-ranked objects from  $O_{\geq k+1}$  in R'' than in R', contradicting  $R' \in \mathcal{R}_{k+1}^{\neq}$ . Any agent  $i \neq j$  who does not rank  $z_{k+1}$  least relative to  $\overline{top}(R') = \overline{top}(R)$ and  $O_{\geq k+1}$ , i.e., for whom  $L(z_{k+1}, R_j) \cap O_{\geq k+1} \cap \overline{top}(R') \neq \emptyset$ , must also rank a below  $z_{k+1}$  in  $R_i = R'_i$ : otherwise, they could move  $z_{k+1}$  to the bottom of their preferences in R' – call the new profile R'''. By BI, we still have  $f_{ja}(R''') < g_{ja}(R''')$ . Again, this would contradict  $R' \in \mathcal{R}_{k+1}^{\neq}$ , i.e., that  $z_{k+1}$  is ranked as low as possible in R'. Since  $a \in L(z_{k+1}, R_j) \cap O_{\geq k+1} \cap \overline{top}(R)$  was chosen arbitrarily, this completes the induction step for Lemma 5.

Induction step for Lemma 6. Suppose the statement is not true for t = k + 1. Then there exist  $R \in \mathcal{R}_{k+1}^{\neq}$  and  $j \in N$  with  $f_{jz_{k+1}}(R) > g_{jz_{k+1}}(R)$ . By Claim 6,  $z_{k+1} \in \overline{top}(R)$ . Moreover, without loss of generality, we may assume that all objects in  $Z_k$  are ranked at the bottom of  $R_j$  such that  $z_{m'}R_jz_m$  for  $m < m' \leq k$ : otherwise we can begin by moving  $z_1$  to the bottom of j's preference list in single, pairwise swaps. Since these transformations keep the profile in  $\mathcal{R}_k^{\neq} \subseteq \mathcal{R}_1^{\neq}$  we have  $f_{jz_1}(\hat{R}) = g_{jz_1}(\hat{R})$  both before and after the swap and hence, by SP,  $f_{jz_{k+1}}(R) > g_{jz_{k+1}}(R)$  (where  $\hat{R}$  denotes an arbitrary profile in the sequence starting at R). Repeating this for each m with  $1 < m \leq k$  establishes the claim.

Consider the partition  $\{I_1, I_2\}$  as in Lemma 5 – by the induction hypothesis and the induction step for Lemma 5 above, this exists for t = k + 1. By Lemma 5 we know that  $j \in I_2$ , i.e.,  $L(z_{k+1}, R_j) \cap O_{\geq k+1} \cap \overline{top}(R) \neq \emptyset$ , and  $0 = \rho_k(z_{k+1}, R) < \rho_j(z_{k+1}, R) \leq \rho_l(z_{k+1}, R)$  for all  $k \in I_1$  and  $l \in I_2$ . As in Lemma 3, we will construct a new profile  $\tilde{R}^*$  in which  $z_{k+1}$  is ranked lower to contradict  $R \in \mathcal{R}_{k+1}^{\neq}$ .

Now, if  $top(R_{I_1}) \cap L(z_{k+1}, R_j) \cap O_{\geq k+1} \neq \emptyset$ , take any  $x \in top(R_{I_1}) \cap L(z_{k+1}, R_j) \cap O_{\geq k+1}$  and move up x in  $R_j$  just below  $z_{k+1}$  to arrive at  $R_j^x$ . Note that  $top(R_j^x, R_{-j}) = top(R)$  and  $L(z_{k+1}, R_j^x) = L(z_{k+1}, R_j)$ . By strategy-proofness, we still have  $f_{jz_{k+1}}(R_j^x, R_{-j}) > g_{jz_{k+1}}(R_j^x, R_{-j})$ . We have either  $f_{jx}(R_j^x, R_{-j}) < g_{jx}(R_j^x, R_{-j})$  or  $f_{jx}(R_j^x, R_{-j}) \ge g_{jx}(R_j^x, R_{-j})$ . We show that for both cases we obtain a new profile R' where  $f_{jz_{k+1}}(R') > g_{jz_1}(R')$ , where  $\rho_i(z_{k+1}, R') = \rho_i(z_{k+1}, R)$  for all  $i \in I_2$  and where  $\rho_i(z_{k+1}, R') \le \rho_j(z_{k+1}, R)$  for all  $i \in I_1$ . Let  $I_1^x$  denote the set of agents in  $I_1$  who rank x at the top.

Case (1.x): if  $f_{jx}(R_j^x, R_{-j}) < g_{jx}(R_j^x, R_{-j})$ , let all  $i \in I_1^x$  push  $\{z_{k+1}\} \cup (L(z_{k+1}, R_j) \cap \overline{top}(R)) \cup Z_k$  to the bottom of their preference order, in the same order as they are ranked in  $R_j$ , to arrive at  $R'_i$ . For j, relabel  $R'_j = R^x_j$  and for all other  $i \in N \setminus (I_1^x \cup \{j\})$ , relabel  $R'_i = R_i$  to arrive at  $R' = (R'_i)_{i \in N}$ . By BI, we still have  $f_{jx}(R') < g_{jx}(R')$ . Towards a contradiction, assume  $f_{jz_{k+1}}(R') \leq g_{jz_{k+1}}(R')$ . Then there would be some object  $yP'_jz_{k+1}$ such that  $f_{jy}(R') > g_{jy}(R')$ . Moreover,  $yP'_iz_{k+1}$  for all  $i \in I_1^x$ . Hence we could push  $z_{k+1}$ down in the preference order, ranking just above  $Z_k$ , for all agents  $i \in I_1^x$  as well as for j and, by BI, arrive at a profile  $\hat{R}$  where f and g differ in the assignment probabilities of y. Since in  $\hat{R}$ ,  $z_{k+1}$  is ranked lower relative to objects  $O_{\geq k+1} \cap \overline{top}(\hat{R}) = O_{\geq k+1} \cap \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_{k+1}^{\neq}$  – and we conclude that  $f_{jz_{k+1}}(R') > g_{jz_{k+1}}(R')$ .

Case (2.x): if instead we have  $f_{jx}(R_j^x, R_{-j}) \ge g_{jx}(R_j^x, R_{-j})$ , swap x and  $z_{k+1}$  in the ranking of j – let us denote this new preference order as  $R'_j$  and the new preference profile  $(R'_j, R_{-j})$ simply as R'. Since  $z_{k+1} \in \overline{top}(R)$ , we have top(R') = top(R) – thus the set of (non-)top ranked objects relevant to determine the ranks of  $z_l$ ,  $l \le k + 1$  in agents preferences is unchanged. Towards a contradiction, assume  $f_{jx}(R') > g_{jx}(R')$ . Then we could push down  $z_{k+1}$  in j's preference order, ranking just above  $Z_k$  and hence below all other  $O_{\ge k+1} \cap$  $\overline{top}(R')$ , and do the same for all  $i \in I_1^x$ , i.e., push down  $\{z_{k+1}\} \cup Z_k$  to the bottom of i's preferences. Call the new preference profile  $\hat{R}$ . By BI the transformation from R' to  $\hat{R}$  preserves  $f_{jx}(\hat{R}) > g_{jx}(\hat{R})$ . Since in  $\hat{R}$  object  $z_{k+1}$  is ranked lower relative to objects  $O_{\ge k+1} \cap \overline{top}(\hat{R}) = O_{\ge k+1} \cap \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_{k+1}^{\neq}$ . Therefore, after having swapped x and  $z_{k+1}$ , we must have  $f_{jx}(R') \le g_{jx}(R')$  and thus  $f_{jz_{k+1}}(R') > g_{jz_{k+1}}(R')$ . Thus, independently of whether Case (1.x) or Case (2.x) applies, we arrive at a new profile R' where  $f_{jz_{k+1}}(R') > g_{jz_{k+1}}(R')$  and where  $\rho_i(z_{k+1}, R') = \rho_i(z_{k+1}, R)$  for all  $i \in I_2$ , i.e.,  $z_{k+1}$  is ranked as low as before for all agents in  $I_2$ . While  $z_{k+1}$  might be ranked higher than in R for agents in  $I_1^x$ , we still have  $\rho_h(z_{k+1}, R') \leq \rho_j(z_{k+1}, R') \leq \rho_l(z_{k+1}, R')$  for all  $h \in I_1$  and  $l \in I_2$ .

Next, if there is any other  $x' \in (top(R_{I_1}) \cap L(z_{k+1}, R_j) \cap O_{\geq k+1}) \setminus \{x\} \subseteq top(R'_{I_1}) \cap L(z_{k+1}, R'_j) \cap O_{\geq k+1}$ , we proceed as before and move up x' in  $R'_j$  just below  $z_{k+1}$ . Refere to this preference order as  $R_j^{x'}$ . By strategy-proofness, we still have  $f_{jz_{k+1}}(R_j^{x'}, R'_{-j}) > g_{jz_{k+1}}(R_j^{x'}, R'_{-j})$ . We proceed as above and obtain profile R'' where  $f_{jz_{k+1}}(R'') > g_{jz_{k+1}}(R'')$  and the rank of  $z_{k+1}$  relative to non-top-ranked objects in  $O_{\geq k+1}$  remains unchanged for agents in  $I_2$ .

Case (1.x'): if  $f_{jx'}(R_j^{x'}, R'_{-j}) < g_{jx'}(R_j^{x'}, R'_{-j})$  we proceed as in Case (1.x) – the only difference is that we now need to take into account the possible changes made to preferences of agents in  $I_1^x$  in Case (1.x). Let all  $i \in I_1^{x'}$  push  $\{z_{k+1}\} \cup (L(z_{k+1}, R_j) \cap \overline{top}(R)) \cup Z_k$  to the bottom of their preference order, in the same order as they are ranked in  $R_j$ , to arrive at  $R''_i$ . For j, relabel  $R''_j = R_j^{x'}$  and for all other  $i \in N \setminus (I_1^{x'} \cup \{j\})$ , relabel  $R''_i = R'_i$  to arrive at  $R'' = (R''_i)_{i \in N}$ . By BI, we still have  $f_{jx'}(R'') < g_{jx'}(R'')$ . Towards a contradiction, assume  $f_{jz_{k+1}}(R'') \leq g_{jz_1}(R'')$ . Then there would be some object  $yP''_jz_{k+1}$  such that  $f_{jy}(R'') > g_{jy}(R'')$ . Moreover,  $yP''_iz_{k+1}$  for all  $i \in I_1^{x'}$  as well as for all  $i \in I_1^x$  if we arrived at R' via Case (1.x). Hence we could push  $z_{k+1}$  down in the preference order, ranking just above  $Z_k$ , for all agents in  $I_1$  for whom we have so far constructed new preferences<sup>34</sup> as well as for j and, by BI, arrive at a profile  $\hat{R}$  where f and g differ in the assignment probabilities of y. Since in  $\hat{R}$ ,  $z_{k+1}$  is ranked lower relative to objects  $O_{\geq k+1} \cap \overline{top}(\hat{R}) = O_{\geq k+1} \cap \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_{k+1}^{\neq}$  – and we conclude that  $f_{jz_{k+1}}(R'') > g_{jz_{k+1}}(R'')$ .

Case (2.x'): if instead we have  $f_{jx'}(R_j^{x'}, R'_{-j}) \geq g_{jx'}(R_j^{x'}, R'_{-j})$ , swap x' and  $z_{k+1}$  in the ranking of j – let us denote this new preference order as  $R''_j$  and the new preference profile  $(R''_j, R'_{-j})$  simply as R''. Since  $z_{k+1} \in \overline{top}(R)$ , we have top(R') = top(R) – thus the set of (non-)top ranked objects relevant to determine the ranks of  $z_l, l \leq k+1$  in agents preferences is unchanged. Towards a contradiction, assume  $f_{jx}(R'') > g_{jx}(R'')$ . Then we could push down  $z_{k+1}$  in j's preference order, ranking just above  $Z_k$  and hence below all other  $O_{\geq k+1} \cap \overline{top}(R')$ , and do the same for all  $i \in I_1^x$ , i.e., push down  $\{z_{k+1}\} \cup Z_k$  to the bottom of i's preferences. Moreover, do the same for all other  $i \in I_1$  for whom we may have so far constructed new preferences. Call the new preference profile  $\hat{R}$ . By BI this preserves  $f_{jx}(\hat{R}) > g_{jx}(\hat{R})$ . Since in  $\hat{R}$  object  $z_{k+1}$  is ranked lower relative to objects  $O_{\geq k+1} \cap \overline{top}(\hat{R}) = O_{\geq k+1} \cap \overline{top}(R)$  than at our initial profile R, this contradicts  $R \in \mathcal{R}_{k+1}^{\neq}$ . Therefore, after having swapped x and

<sup>&</sup>lt;sup>34</sup>I.e., for  $i \in I_1^{x'} \cup I_1^x$  if we arrived at R' via Case (1.x), and for  $i \in I_1^{x'}$  if we arrived at R via Case (2.x).

z, we must have  $f_{jx}(R'') \leq g_{jx}(R'')$  and thus  $f_{jz_{k+1}}(R'') > g_{jz_{k+1}}(R'')$ .

Repeat these steps for all  $x^* \in top(R_{I_1}) \cap L(z_{k+1}, R_j) \cap O_{\geq k+1}$ , i.e., move up  $x^*$  in the preference order of j to just below  $z_{k+1}$  and then proceed as in Case (1.x') or (2.x'). This way, we arrive at a profile, refer to it as  $R^{\dagger}$ , where  $top(R_i^{\dagger}) = top(R_i)$  for all  $i \in N$ ,  $f_{jz_{k+1}}(R^{\dagger}) > g_{jz_{k+1}}(R^{\dagger})$ , and  $I_1$  has been partitioned into two subsets:  $I'_1$  includes all agents  $i \in I_1$  for whom  $R_i^{\dagger} = R_i$  and hence  $L(z_{k+1}, R_i^{\dagger}) \cap O_{\geq k+1} \cap \overline{top}(R) = \emptyset$ , and whose top ranked objects are in  $Z_k$  or ranked above  $z_{k+1}$  by j - in R but also in  $R^{\dagger}$  since j's lower contour set has only gotten weakly smaller as we moved away from R to  $R^{\dagger}$  (strictly smaller whenever Case 2 applied). Second,  $I''_1$  includes all agents  $i \in I_1$  whose lower contour set  $L(z_{k+1}, R_i^{\dagger})$ consists of all objects  $(L(z_{k+1}, R_j) \cap \overline{top}(R)) \cup Z_k$ . Third, compared to R, j's lower contour set at  $z_{k+1}$  has gotten weakly smaller in that some objects from  $top(R_{I_1})$  may now be ranked above  $z_{k+1} -$  however, no object in  $O_{\geq k+1} \cap \overline{top}(R)$  has been raised above  $z_{k+1}$  as we moved to  $R_j^{\dagger}$ , i.e.,  $L(z_{k+1}, R_j) \cap O_{\geq k+1} \cap \overline{top}(R) = L(z_{k+1}, R_j^{\dagger}) \cap O_{\geq k+1} \cap \overline{top}(R^{\dagger})$ . Last the ranking of other agents  $i \in I_2 \setminus \{j\}$  is unchanged, i.e.,  $R_i^{\dagger} = R_i$ .

By the induction step for Lemma 5 as well as the preceding construction, we have for all  $h \in I_1''$  and all  $l \in I_2$ ,

$$L(z_{k+1}, R_h^{\dagger}) \cap O_{\geq k+1} \cap \overline{top}(R^{\dagger}) \subseteq L(z_{k+1}, R_j^{\dagger}) \cap O_{\geq k+1} \cap \overline{top}(R^{\dagger}) \subseteq L(z_{k+1}, R_l^{\dagger}) \cap O_{\geq k+1} \cap \overline{top}(R^{\dagger})$$
  
$$\rho_h(z_{k+1}, R^{\dagger}) \leq \rho_j(z_{k+1}, R^{\dagger}) \leq \rho_l(z_{k+1}, R^{\dagger}) \text{ and } \rho_l(z_{k+1}, R^{\dagger}) = \rho_l(z_{k+1}, R).$$

Now, for all  $i \in I_1'' \cup I_2$  (including j) change the order of objects in the lower contour set  $L(z_{k+1}, R_i^{\dagger})$  as follows: (i) objects that are in  $L(z_{k+1}, R_i^{\dagger}) \setminus L(z_{k+1}, R_j^{\dagger})$  are ranked immediately below  $z_{k+1}$  (beyond that, their order does not matter), (ii) objects that are also in  $L(z_{k+1}, R_j^{\dagger}) \cap \overline{top}(R^{\dagger})$  are ranked next, in the same order as by  $R_j^{\dagger}$ , (iii) last, all objects in  $L(z_{k+1}, R_i^{\dagger}) \cap L(z_{k+1}, R_j^{\dagger}) \cap top(R^{\dagger})$  are ranked below (beyond that, their order does not matter). Call this new (and penultimate) profile  $\tilde{R}$ . By BI, we still have  $f_{jz_{k+1}}(\tilde{R}) > g_{jz_{k+1}}(\tilde{R})$ . By Lemma 2 and  $f \triangleright^{sd} g$ , we have  $f_{ix}(\tilde{R}) = 0 = g_{ix}(R)$  for all  $i \in I_1'' \cup I_2$  and all  $x \in L(z_{k+1}, \tilde{R}_i) \cap L(z_{k+1}, \tilde{R}_j) \cap top(\tilde{R})$ . Lastly, since in moving from R to  $R^{\dagger}$  and on to  $\tilde{R}$  objects in  $Z_k$  were only moved down relative to non-top ranked objects, we still have  $\tilde{R} \in \mathcal{R}_k^{\neq}$ .

Hence, we now have all agents in  $I''_1 \cup I_2$  ranking objects  $L(z_{k+1}, \tilde{R}_j) \cap \overline{top}(\tilde{R})$  adjacent and in the same order as  $\tilde{R}_j$ , and below that only objects in  $top(\tilde{R}) = top(R)$  for which the assignment probabilities are equal to zero under f and g by Lemma 2. Since  $f_{jz_{k+1}}(\tilde{R}) >$  $g_{jz_{k+1}}(\tilde{R})$ , there is some y, ranked below  $z_{k+1}$  by  $\tilde{R}_j$ , such that  $f_{jy}(\tilde{R}) < g_{jy}(\tilde{R})$  – and thus some  $i \in N$  with  $f_{iy}(\tilde{R}) > g_{iy}(\tilde{R})$ . Moreover, by Lemma 2 and the induction hypothesis, we have  $y \in L(z_{k+1}, \tilde{R}_j) \cap O_{\geq k+1} \cap \overline{top}(\tilde{R})$ . If  $i \in I_1'' \cup I_2$ , then there is  $y' \neq y$  with  $y\tilde{R}_iy'$ , such that  $f_{iy'}(\tilde{R}) < g_{iy'}(\tilde{R})$  – and thus some  $i' \in N$  with  $f_{i'y'}(\tilde{R}) > g_{i'y'}(\tilde{R})$ . Moreover, given that  $y \in L(z_{k+1}, \tilde{R}_j)$ , our construction of  $\tilde{R}_i$  implies that y' is ranked lower than y according to  $\tilde{R}_j$ , while by Lemma 2 and the induction hypothesis, it must be that  $y' \in L(y, \tilde{R}_j) \cap O_{\geq k+1} \cap \overline{top}(\tilde{R})$ .

If  $i' \in I_1'' \cup I_2$ , then there is  $y'' \neq y'$  with  $y'\tilde{R}_{i'}y''$ , such that  $f_{i'y''}(\tilde{R}) < g_{i'y''}(\tilde{R})$  – and thus some  $i'' \in N$  with  $f_{i''y''}(\tilde{R}) > g_{i''y''}(\tilde{R})$ , and so on.

Since  $L(z_{k+1}, \tilde{R}_j) \cap O_{\geq k+1} \cap \overline{top}(\tilde{R})$  is finite and we move down (according to  $\tilde{R}_j$ ) in each iteration, eventually there is some  $y^* \in L(z, \tilde{R}_j) \cap O_{\geq k+1} \cap \overline{top}(\tilde{R})$  and  $i^* \in I'_1 = N \setminus (I''_1 \cup I_2)$  such that  $f_{i^*y^*}(\tilde{R}) > g_{i^*y^*}(\tilde{R})$ .

Suppose  $top(\tilde{R}_{i^*}) \in O_{\geq k+1}$ . Note that  $\tilde{R}_i = R_i$ , and thus,  $y^*\tilde{P}_i z_1$  for any  $i \in I'_1$ . For any  $i \in I'_1$  where  $y^*P_i top(\tilde{R}_{i^*})$ , change  $\tilde{R}_i$  to  $\tilde{R}'_i$  as follows: (i) objects in  $B(y^*, R_i)$  are ranked first according to  $R_i$ , (ii) then  $top(\tilde{R}_{i^*})$  and (iii) then objects in  $L(y^*, R_i) \setminus \{top(\tilde{R}_{i^*})\}$ according to  $R_i$ . After having done this for all such  $i \in I'_1$  and denoting the obtained profile by  $\tilde{R}'$ , by BI we continue to have  $f_{i^*y^*}(\tilde{R}') > g_{i^*y^*}(\tilde{R}')$ . As only  $top(\tilde{R}_{i^*})$  was moved up and  $\tilde{R} \in \mathcal{R}_k^{\neq}$ , we still have  $\tilde{R}' \in \mathcal{R}_k^{\neq}$ . But then let  $i^*$  exchange the positions of  $y^*$  and  $top(\tilde{R}_{i^*})$  in  $\tilde{R}'_{i^*}$  and call this final profile  $\tilde{R}^*$ . This strictly decreases the number of non-top objects ranked below  $z_{k+1}$  for j, as well as all  $i \in I''_1$ , and weakly decreases it for all  $i \in I'_1$ (as either  $top(\tilde{R}_{i^*})\tilde{P}'_iy^*\tilde{P}'_iz_{k+1}$  or  $top(\tilde{R}_{i^*})$  is ranked immediately below  $y^*$  in  $\tilde{R}'_i$ ) and for all  $i \in I_2 \setminus \{j\}$  (only weakly if  $i \in I_2 \setminus \{j\}$  ranked both  $top(\tilde{R}_{i^*})$  and  $top(\tilde{R}_{i^*})$  below  $z_{k+1}$ ). Note that this also weakly decreases the number of non-top objects ranked below any object in  $Z_k$ . Hence,  $\rho_i(z_{k+1}, \tilde{R}^*) \leq \rho_i(z_{k+1}, R)$  for  $i \in I_2 \setminus \{j\}$ ,  $\rho_j(z_{k+1}, \tilde{R}^*) < \rho_j(z_{k+1}, R)$ , and  $\rho_i(z_{k+1}, \tilde{R}^*) \leq \rho_j(z_{k+1}, \tilde{R}^*)$  for  $i \in I_1$  contradicting  $R \in \mathcal{R}_{k+1}^{\neq}$ .

Finally, consider  $top(\tilde{R}_{i^*}) = z_m \in Z_k$  (i.e.,  $m \leq k$ ). Since  $f_{i^*y^*}(\tilde{R}) > g_{i^*y^*}(\tilde{R})$  there must be some lower ranked  $\hat{y}$  such that  $f_{i^*\hat{y}}(\tilde{R}) < g_{i^*\hat{y}}(\tilde{R})$ . But then, consider the strict upper countour set of  $\hat{y}$ , i.e.  $U(\hat{y}, \tilde{R}_i^*) = \{o \in O : o\tilde{P}_i^*\hat{y}\}$ . Push all elements in  $U(\hat{y}, \tilde{R}_{i^*}) \cap Z_k$  to just above  $\hat{y}$  to arrive at  $\tilde{R}^*$ . This preserves  $f_{i^*\hat{y}}(\tilde{R}^*) < g_{i^*\hat{y}}(\tilde{R}^*)$  (by SP). Moreover, since we have pushed these objects below  $y^*$  and  $y^* \in O_{\geq k+1} \cap \overline{top}(\tilde{R})$ , we have reduced their rank. But that contradicts  $\tilde{R} \in \mathcal{R}_k^{\neq}$  – which concludes the proof.

### C Independence of Non-Bossiness in Theorem 3

Using the example from Svensson (1999, p.563) we show that strategy-proofness, efficiency and bounded invariance do not imply non-bossiness.

**Example 1** Let  $N = \{1, 2, 3, 4\}$  and  $O = \{a, b, c, d\}$ .

Let g be defined as follows: if  $R_1 = R_2$ , then use the SD order 1 - 2 - 3 - 4, and

otherwise use the SD order 1 - 2 - 4 - 3. It is easy to see that g satisfies strategy-proofness and efficiency, and violates non-bossiness. We show that g satisfies bounded invariance.

Let R be a profile. Agents 3 and 4 cannot affect who gets objects in their upper contour set by changing their ranking below, since from their perspective the ordering in which agents choose is exogenous – their lower contour set for a given object only matters if all objects in their upper contour set have already been assigned to other agents by the time they get to pick an object.

Now for agents 1 and 2 there are only two ways in which an agent can change their ranking such that bounded invariance has any bite – either they change the ranking below their top ranked object (this should not affect who gets their top-ranked object) or they change the order of the two least preferred objects (this should not change who gets the two most-preferred objects)

Let denote the top ranked object of agent 1 by x and consider three cases.

Case 1: Both 1 and 2 rank x at the top and the same object at the second position:  $R_1 : xy \dots$ and  $R_2 : xy \dots$ 

If 1 changes the ordering below x, she still gets x; if 1 changes the order of the last two objects, 2 still gets y. The same holds for 2.

Case 2: Both 1 and 2 rank x at the top but different objects at second position:  $R_1: xy...$  and  $R_2: xz...$ 

If 1 changes the ordering below x, she still gets x; if 1 changes the order of the last two objects, 2 still gets z and the same SD order 1 - 2 - 4 - 3 is used. Same holds for 2.

Case 3: 1 and 2 rank different objects at the top:  $R_1 : x \dots$  and  $R_2 : y \dots$ 

No matter whether 1 or 2 change the order below their most-(or second-most-)preferred object, 1 receives x and 2 receives y while 4 gets to choose next and receives her most preferred object among  $O \setminus \{x, y\}$ .